

Water Resources Research[®]

RESEARCH ARTICLE

10.1029/2022WR034053

Key Points:

- Integrating groundwater data and streambed micro-topography can improve the interpretation of transient storage model results
- Parameter interaction in transient storage models increases with discharge
- Hyporheic and in-stream storage can both contribute to increase and decrease of stream water transient storage when discharges increase

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

E. Bonanno, bonanno@hydro.tuwien.ac.at

Citation:

Bonanno, E., Blöschl, G., & Klaus, J. (2023). Discharge, groundwater gradients, and streambed micro-topography control the temporal dynamics of transient storage in a headwater reach. *Water Resources Research*, 59, e2022WR034053. https:// doi.org/10.1029/2022WR034053

Received 6 NOV 2022 Accepted 17 JUN 2023

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Discharge, Groundwater Gradients, and Streambed Micro-Topography Control the Temporal Dynamics of Transient Storage in a Headwater Reach

Enrico Bonanno^{1,2} , Günter Blöschl², and Julian Klaus³

¹Catchment and Eco-Hydrology Group, Luxembourg Institute of Science and Technology, Belvaux, Luxembourg, ²Institute of Hydraulic and Water Resources Engineering, Vienna University of Technology, Vienna, Austria, ³Institute of Geography, University of Bonn, Bonn, Germany

Abstract Contradictory interpretations of transient storage modeling (TSM) results of past studies hamper the understanding of how hydrologic conditions control solute transport in streams. To address this issue, we conduct 30 instantaneous tracer experiments in the Weierbach stream, Luxembourg. Using an iterative modeling approach, we calibrate TSM parameters and assess their identifiability across various hydrologic conditions. Near-stream groundwater monitoring wells and LIDAR scans of the streambed are used to evaluate the area of the hyporheic zone and of the submerged sediments for each experiment. Our findings show that increasing discharge enhances parameters interaction requiring more samples of TSM parameters to obtain identifiable results. Our results also indicate that transient storage at the study site is influenced by in-stream and hyporheic exchange processes during low discharge, likely due to the hyporheic zone's large extent and the relatively low water level compared to the size of slate fragments on the streambed. However, as discharge increases, in-stream storage zones become part of the advective channel and the lower localized stream water losses to the adjacent groundwater suggests a decrease of the hyporheic exchange on transient storage. The results obtained were utilized to generate a hydrograph for the study site illustrating the dynamic evolution of in-stream and hyporheic storage with varying discharge, providing insights into the expected influence of different transient storage processes prior to tracer experiments. Overall, our study enhances the understanding of the role of the hyporheic area and in-stream storage zones in transient storage and helps estimate TSM parameters more accurately.

Plain Language Summary The transport of substances in stream networks is not only controlled by the velocity of the water, but also by the shape of the stream's channels and the continuous exchange of water between the stream and the saturated soil adjacent to its banks, that is, groundwater. A common approach to studying water movement in streams is to release a tracer—a substance that moves like water—into the stream water and measure its transport downstream. Although this method has provided useful results in the past, understanding the processes that control water transport and their changes over time is still challenging. Here, we studied the transport of a tracer in a stream reach at different times of the year and under varying flow conditions. We found that the exchange of water between the stream and groundwater, together with the slow movement of water between the stones in the stream channel increased, water velocity and eddies generated by the friction with the streambed sediments became the predominant factor in controlling downstream water transport, and the exchange between the stream channel and adjacent groundwater became less important.

1. Introduction

The understanding of how different hydrological processes control the transport of water downstream remains one of the most challenging topics in stream hydrology (Boano et al., 2014; Kelleher et al., 2019). The movement of water downstream is not only controlled by the advection-dispersion process, but it is also governed by the transient storage process (Gooseff, Bencala, et al., 2008). Transient storage processes are the combined result of different mechanisms that delay the transport of water and solutes downstream compared to the advection-dispersion processes. Transient storage can occur due to slowly moving surface water within the channel (i.e., in-stream storage zones, Nordin & Troutman, 1980), due to the hydrologic exchange with subsurface water adjacent to and



underlying the stream channel (i.e., the hyporheic zone, Bencala & Walters, 1983; Cardenas & Wilson, 2007), and due to the development of eddies and turbulences in the water column resulting from friction with in-stream obstructions (Jackson et al., 2013). Despite the amount of work investigating the role of different processes on the transient storage of water in streams, the current state-of-the-art is ambiguous on the understanding of how different hydrological processes and their interaction contribute to the transient storage across spatial scales and hydrologic conditions. Transient storage models (TSMs) are phenomenological models that provide a valuable tool to distinguish the role of advection-dispersion and transient storage processes on water and solute transport in stream channels (Butturini & Sabater, 1999; Jin & Ward, 2005; Morrice et al., 1997; Zarnetske et al., 2007). However, in most studies the role of in-stream storage zones and the hyporheic zone is inferred from TSM results only for a few hydrologic conditions and without information or data that can inform on the effective presence of transient storage zones (Runkel et al., 2003; Ward et al., 2019; Ward & Packman, 2019; Wondzell, 2006). These limitations hinder our ability to decipher how different parameters in the TSM influence the modeled BTC over the hydrological year and which hydrological process they indicate (Knapp & Kelleher, 2020).

The need to understand how different hydrological processes can influence water chemistry, biological activity, and the ecological richness of stream networks has motivated a range of studies over the past 30 years (Boulton et al., 2010; Stanford & Ward, 1988; Ward, 2016). Here, TSMs have been adopted to characterize hyporheic exchange and in-stream water storage in a multitude of streams (Bencala et al., 2011; Bencala & Walters, 1983; Butturini & Sabater, 1999; Gooseff et al., 2005, 2008; Hart et al., 2002; Ward et al., 2018; Wörman, 1998). TSM assumes a uniform, steady-state, 1-D flow modeled via the advection-dispersion equation (ADE) with a first-order mass transfer exchange α between the advective flow channel and a finite-sized storage zone with $A_{\rm TS}$ dimensions (Bencala & Walters, 1983). Knowing how the transient storage parameters α and $A_{\rm TS}$ change with different hydrologic conditions is crucial as larger transient storage zones and longer residence times are key factors in enhancing nutrient cycling (Argerich et al., 2011) and degrading pollutants (Moser et al., 2003) in stream networks. The simplified, yet informative structure of the TSM can thus offer valuable insights into the potential development of hot spots and hot moments that control water quality in surface waters (Krause et al., 2017; Smith, 2005).

Despite the pressing need to decipher the role of different hydrological processes and conditions on solute transport in streams, current research led to a collection of idiosyncratic studies with conflicting model predictions and interpretations (Ward & Packman, 2019). Previous studies have primarily investigated the variation of TSM parameters in terms of spatial relationships, interpreting changes in discharge as a function of increasing reach length or multiple adjacent reaches (Bencala & Walters, 1983; Bencala et al., 1990; D'Angelo et al., 1993; Scott et al., 2003; McKnight et al., 2004; Ward, Gooseff, et al., 2013; Gooseff et al., 2013). Alternatively, other studies have focused on changes in discharge and TSM parameters within the same stream reach (Edwardson et al., 2003; Jin & Ward, 2005; Mason et al., 2012; Schmid et al., 2010; Ward et al., 2018; Wlostowski et al., 2017). Despite the different approaches, no study has yet focused on the change of TSM parameters with discharge over the same stream reach for a wide range of hydrologic conditions (Table 1).

Higher discharge has been linked to higher (Bencala & Walters, 1983; Lees et al., 2000; Schmid et al., 2010; Ward et al., 2018; Wlostowski et al., 2013) and lower (Harvey et al., 1996; Karwan & Saiers, 2009; Martí et al., 1997; Morrice et al., 1997; Valett et al., 1996; Ward et al., 2017; Wondzell, 2006) values of the transient storage area A_{TS} , or to show no clear relationships (Edwardson et al., 2003; Fabian et al., 2011; Hall et al., 2002; Jin & Ward, 2005; McKnight et al., 2004; Ward, Payn, et al., 2013; Zarnetske et al., 2007). Similarly, the rate of exchange α between the advective flow channel with the transient storage zone was higher (Fabian et al., 2011; Gooseff et al., 2003; Hart et al., 1999; Harvey et al., 1996; Lees et al., 2000; Schmid et al., 2010; Ward et al., 2018; Wondzell, 2006), lower (González-pinzón et al., 2015; Karwan & Saiers, 2009; Ward et al., 2017), or showed no clear relationship with discharge (Bencala et al., 1990; Bencala & Walters, 1983; Butturini & Sabater, 1999; D'Angelo et al., 1993; Edwardson et al., 2003; Gooseff et al., 2013; Hall et al., 2002; Jin & Ward, 2005; Legrand-Marcq & Laudelout, 1985; Martí et al., 1997; McKnight et al., 2004; Morrice et al., 1997; Scott et al., 2003; Ward, Payn, et al., 2013).

The inconsistency in the relation between the values of TSM parameters with discharge in previous studies (Table 1) might derive from the specific characteristics of the different study sites. For example, stream channels with relatively high hydraulic conductivity of the streambed material (sand and gravel) would allow the inflow of water from the stream toward the adjacent groundwater with higher discharge (Dudley-Southern & Binley, 2015),

	2	Number and/or name	Number of	A _{TS} versus	a versus		
Study	Site name	of the sub-reaches	experiments per reach	б	б	Fitting	Identifiability analysis
Legrand-Marcq and Laudelout (1985)	Rieu d'Ostenne	1	13	()	Unclear	Visual No	0
Bencala and Walters (1983)	Uvas Creek	Ś	1	÷	Unclear	Visual Nc	0
Bencala et al. (1990)	Snake river (upstream Deer Creek)	2	1	Unclear	Unclear	Visual	0
	Snake river (downstream Deer Creek)	ς	-	Unclear	Unclear	Visual Nc	0
D'Angelo et al. (1993)	Artificial streams	2 (Dogwood and Oak)	2 (summer-winter)	~	~	Visual	0
	First-order site	2 (Pine and Hardwood)	2 (summer-winter)	÷	(+)		
	Gradient site	4 (from first to fourth order)	1 (summer)	Unclear	Unclear		
	Fifth-order site—unconstrained	2 (reaches 4 and 7)	2 (summer-winter)	Ĵ	Unclear		
	Fifth-order site—constrained	2 (reaches 1, 2, 3, 5, 6)	1 (summer)	Unclear	Unclear		
Valett et al. (1996)	Gallina Creek	1	4	-	/	Visual	0
Harvey et al. (1996)	St. Kevin Gulch	1	2	-	(+	Nonlinear least squares regression	0
Morrice et al. (1997)	Gallina Creek	1	4	Ĵ	Unclear	Visual	0
Martí et al. (1997)	Sycamore Creek	1	8	Ĵ	Unclear	Visual	0
Butturini and Sabater (1999)	Riera Major stream	Π	S	Unclear	Unclear	Direct fitting (sensu Hart, 1995) No	0
Hart et al. (1999)	West Fork of Walker Branch	Π	11 (³ H) and 9 (Cl ⁻)	Unclear	(+)	Direct fitting (sensu Hart, 1995) No	0
Lees et al. (2000)	Mimram River	0	-	÷	()	OTIS-P (sensu Runkel, 1998) NG	on-unique convergence of α and A_{TS} parameters.



10.1029/2022WR034053



Table 1Continued							
Study	Site name	Number and/or name of the sub-reaches	Number of experiments per reach	$A_{ m TS}$ versus Q	α versus Q	Fitting	Identifiability analysis
Hall et al. (2002)	Hubbard Brook	1 (Bear brook)	4	Unclear	Unclear	Direct fitting (sensu Hart, 1995)	No
	Experimental Forest	1 (Cone Pond Outlet)	0	(+)	÷		
		1 (Hubbard brook)	3	Unclear	Unclear		
		1 (Paradise brook)	2	(+)	()		
		W2 stream	33	(+)	Unclear		
		W3 stream	6	Unclear	Unclear		
		W4 stream	3	Unclear	Unclear		
		W5	3	(+)	Unclear		
		W6	5	Unclear	Unclear		
		West inlet to mirror lake	£	Unclear	Unclear		
Edwardson et al., 2003	Imnavait Creek	Site 1	2	(+)	Ĵ	OTIS-P (sensu Runkel, 1998)	No
		Site 2	7	-	(+)		
	Blueberry Creek	Site 1	4	Unclear	(+)		
		Site 2	4	Unclear	Unclear		
	Toolik Inlet Stream	Site 1	2	+	+		
	Oksrukuyik Creek	Site 1	2	1	+		
	Kuparuk River	Site 1	6	Unclear	Unclear		
		Site 2	9	Unclear	Unclear		
Gooseff et al. (2003)	Lookout Creek watershed	L0411	6	Ĵ	(+)	UCODE plus manual modification of the parameters to visually match the tail of the BTC.	No
Scott et al. (2003)	Uvas Creek	Ś	-	Ĵ	Unclear	UCODE (unable to calibrate α and $A_{\rm TS}$ for two over five BTCs)	No
McKnight et al. (2004)	McMurdo Dry Valleys (Green Creek)	4	-	Unclear	Unclear	OTIS-P (sensu Runkel, 1998)	No
Jin and Ward (2005)	Payne Creek	1	9 (constant-rate)	Unclear	Unclear	OTIS-P (sensu Runkel, 1998)	No
			6 (slug)	Unclear	Unclear		
Wondzell (2006)	WS1	1 (upper)	7	1	()	OTIS-P (sensu Runkel, 1998)	No
		1 (lower)	7	<u>(</u>)	(+		
	WS3	1 (upper)	2	()	<u> </u>		
		1 (lower)	2	1	(+)		

Study	Site name	Number and/or name of the sub-reaches	Number of experiments per reach	$A_{ m TS}$ versus Q	α versus Q	Fitting	Identifiability analysis
Zarnetske	Northern foothills of	1 (A2)	3	(+)	÷	STAMMT-L (sensu Haggerty et al., 2002)	No
et al. (2007)	Alaska's Brooks	1 (P1)	4	Unclear	()		
	Naligo	1 (AP)	4	Unclear	(+)		
Karwan and Saiers (2009)	Wangum Brook	1	3 (2)	Ĵ	Ĵ	Levenberg-Marquardt nonlinear least squares algorithm (unable to calibrate α and $A_{\rm TS}$ under high discharge stages)	No
Schmid et al. (2010)	Mödlingbach	1	12	(+)	(+)	OTIS-P (sensu Runkel, 1998)	No
	Torrente Lura	S-I	4	(+)	(+)		
		S-II	7	(+)	(+)		
		D-G	5	(+)	(+)		
Fabian et al. (2011)	Prieta Creek	1	3	Unclear	(+)	OTIS-P (sensu Runkel, 1998)	No
Mason et al. (2012)	Silver Bow Creek	1	58	~	~	1	Identifiability analysis— α and $A_{\rm TS}$ non-identifiable
Ward, Gooseff, et al. (2013)	Stringer Creek	28	4	Unclear	Unclear	OTIS + Monte Carlo simulations (100,000)	Identifiability analysis— α and $A_{\rm TS}$ non-identifiable
Gooseff et al. (2013)	Uvas Creek	5 (+10 sub-reach combinations)	-	(+)	Unclear	OTIS + UCODE (sensu Scott et al., 2003)	No (suspected non-identifiability for some results)
González-pinzón et al. (2015)	Shaver Creek	6	1	÷	()	OTIS + Shuffled Complex Evolutionary algorithm	No
Wlostowski et al. (2017)	Alaska's North Slope	I8 inlet Peat inlet	4 ε	Unclear (+)	Unclear (+)	OTIS + Shuffled Complex Evolutionary algorithm	No
Ward et al. (2017)	Tenderfoot Creek Experimental Forest	2 (100–2,500 m)	1	1	Ĺ	OTIS + Monte Carlo simulations (100,000)	Identifiability analysis— D , A , α and A_{TS} non-identifiable
Ward et al. (2018)	Fawn River	Unrestored reach	4	(+	()	OTIS + Monte Carlo simulations (100,000)	Identifiability analysis— D , A , α and A_{TS} identifiable
<i>Note.</i> (+) and (–) sym and negative relationsl investigated multiple s was studied via multip	bols indicate respectively p hips have been observed be tream reaches, we reported ole reaches with tracer meas	ositive and negative relative the discharge and α a only the reaches investigant entrement location locate	tionships between stream nd $A_{\rm TS}$. The slash symbo gated at different hydrolc d at increasing distance	n discharge ol (/) indicat ogic conditio from the tra	with the tr es no resu ons. We he cer injecti	ansient storage parameters α and A_{1S} . The "unclea lts, due to non-reported results or a null evaluation are also reported studies using only one tracer injec on point.	r" term means that both positive to f the parameter. When a study tion, but where higher discharge

ADVANCING EARTH AND SPACE SCIENCES leading to a more pronounced tail of the BTC (Schmadel et al., 2016). In contrast, stream channels confined by fresh bedrock or characterized by material with low hydraulic conductivity would exhibit a rather reduced hyporheic zone area at high discharge and are thus more likely to show a reduction in transient storage area with higher discharge (Wondzell, 2011). Nevertheless, conflicting or absent relationships between TSM parameters and stream discharge could also be due to limitations that are common in studies investigating solute transport in streams with tracer experiments (Table 1). First, the calibration of parameters in TSM has been performed iteratively to visually fit the modeled BTC over the observed BTC (Bencala et al., 1990; Bencala & Walters, 1983; D'Angelo et al., 1993; Legrand-Marcq & Laudelout, 1985; Valett et al., 1996), or by inverse modeling (Edwardson et al., 2003; Fabian et al., 2011; Gooseff et al., 2013; Jin & Ward, 2005; Lees et al., 2000; McKnight et al., 2004; Schmid et al., 2010; Wondzell, 2006). However, the identifiability of model parameters has generally not been taken into account in most TSM studies (Table 1), leading to a lack of certainty about the modeling results and their physical interpretation (Knapp & Kelleher, 2020). When TSM parameters are non-identifiable, they are highly interdependent, meaning that changes in one parameter would be balanced by a proportional change of one or more other parameters leading to the same model performances (Camacho & González, 2008; Kelleher et al., 2013; Wagener, Lees, & Wheater, 2002; Wagner & Harvey, 1997). Identifiability is a crucial issue for the interpretation of TSM results, as most studies that have addressed the identifiability of TSM parameters have found that they were non-identifiable (Camacho & González, 2008; Kelleher et al., 2019; Wagener, Lees, & Wheater, 2002; Ward, Kelleher, et al., 2017; Wlostowski et al., 2013). Non-identifiability of TSM parameters does not only affect model performance, but it can result in the modeled BTC mimicking the advection-dispersion equation, leading to a misinterpretation of the processes governing transient storage at the study site (Bonanno, Blöschl, & Klaus, 2022). A growing number of studies addressed parameter identifiability in TSMs via random sampling approaches (Kelleher et al., 2019; Ward et al., 2017, 2018; Knapp & Kelleher, 2020; Table 1). However, no study to date has directly investigated the identifiability of the TSM parameters under multiple hydrologic conditions, which may improve our understanding of why TSM parameters were identifiable in some studies (Ward et al., 2017, 2018) and not in others (Camacho & González, 2008; Kelleher et al., 2013; Wagener, Lees, & Wheater, 2002; Ward, Kelleher, et al., 2017; Wlostowski et al., 2013).

There is also a second limitation that may cause an unclear correlation between TSM parameters and discharge, thus hindering the physical interpretation of the model results, namely the scarcity of information about the stream reach and the relatively small number of tracer experiments. Studies analyzing a section of a stream under different hydrological conditions, or studying higher stream discharge at successive monitoring stations, rarely investigated more than four stages of discharge (Table 1) with the tracer experiments being mostly conducted at baseflow conditions (Ward, 2016). A relatively low number of investigated discharge stages hampers the ability to observe a robust relationship between discharge and α and A_{TS} , resulting in a poor understanding of the processes controlling transient storage in stream reaches (Ward & Packman, 2019). Investigating the link between TSM parameters across different discharge stages is also not enough for associating specific hydrological processes with certain hydrologic conditions. This is because higher discharge can cause larger and lower hyporheic exchange and in-stream transient storage depending on the stream morphology and the groundwater gradients at the study site (Bonanno et al., 2021; Dudley-Southern & Binley, 2015; Gooseff, Bencala, et al., 2008; Jin & Ward, 2005; Martí et al., 1997; Schmid et al., 2010). A clearer perception of the physical processes of stream reaches fosters a robust interpretation of TSM results. As an example, the measurements of the groundwater levels adjacent to the stream channel can be used to infer if the near-stream groundwater is receiving stream water, thus offering valuable information on the potential development of hyporheic zones at the study site (Bonanno et al., 2021; Voltz et al., 2013; Wondzell, 2006). Likewise, the size and distribution of the sediments making the streambed topography can be a crucial resource for the interpretation of TSM results, since streambed sediments and in-stream obstructions can create recirculating zones resulting in a non-negligible role for in-stream transient storage (Hart et al., 1999; Jackson et al., 2013; Montgomery & Buffington, 1997). Without a comprehensive knowledge of the groundwater levels and the streambed topography, it is difficult to distinguish whether a certain transient storage area can be related to a specific transient storage process or whether it is simply the result of a mathematical fit of the TSM parameters without a realistic role for solute transport at the study site.

In this manuscript we address the following research questions.

- 1. Does the identifiability of TSM parameters change with discharge?
- 2. How do transient storage processes change under different hydrologic conditions?





Figure 1. (a) Study reach and location of the wells and piezometers. The *Y* direction corresponds to the direction of the geographic north; thus the stream reach defines an "east footslope" on the right side of the map, and a "west footslope," on the left side of the map. (b) Transects of the stream channel were extracted from four LIDAR scans at a 1 cm resolution and 15 cm distance between each other. The white areas between the scans are missing data due to the local presence of water and shadow zones. Blue, green, and red lines indicate the length of the wetted perimeter in each transect for the three reported discharge stages. (c) Flow duration curve for the recorded discharge at the Weierbach outlet for the period January 2018—December 2021, 55 m upstream of the study site. (d) Hydrograph of the Weierbach outlet for the investigation period. The vertical red lines in (c) and (d) indicate the discharge stages investigated during the slug tracer injections.

To answer these questions and to overcome the limitations mentioned above in TSM studies we performed 30 in-stream tracer experiments and we: (a) investigated the identifiability of TSM parameters by combining global identifiability analysis with dynamic identifiability analysis in an iterative approach obtaining identifiable TSM parameters (Bonanno, 2022; Bonanno, Blöschl, & Klaus, 2022); (b) we recorded the groundwater elevation at the study site through a groundwater monitoring network of 43 wells that allowed us to infer the extent of the hyporheic zone during each experiment; (c) we obtained the micro-topography of the streambed via a laser scan and we compared the distribution of the height of the slate fragments in the streambed to the surface water level to infer the role of in-stream transient storage in different hydrologic conditions.

2. Methods

2.1. Study Site

The study reach is located in western Luxembourg, downstream of the Weierbach experimental catchment ($49^{\circ}49'38''N$, $5^{\circ}47'44''E$) (Hissler et al., 2021). The stream reach is 55 m long and it is characterized by a riffle morphology, has an average slope of $\simeq 6\%$, is unvegetated, and consists of deposited colluvium of fragmented slates over a fractured bedrock layer (Bonanno et al., 2021; Figure 1). Previous work outlined the occurrence of several hydrological processes controlling stream water generation in the Weierbach catchment. The hillslopes at the study site are characterized by a regolith layer with a relatively high hydraulic conductivity compared to the fractured bedrock layer beneath (Glaser et al., 2016, 2020). The subsurface structure does not promote shallow lateral flow toward the stream channel (Klaus & Jackson, 2018), and precipitation water percolates vertically toward the groundwater table in the fractured bedrock (Rodriguez & Klaus, 2019). The water movement through and above the fractured bedrock from the hillslopes maintains a rather steady and shallow groundwater level in the near-stream domain throughout the year (Fabiani et al., 2021), and the organic soil areas composing part of the riparian area are almost constantly saturated (Bonanno et al., 2021). Discharge is thus generated by both a fast-

Table 2

List of the In-Stream Instantaneous Tracer Injections, Date, Discharge From Dilution Gauging, and Amount of Injected NaCl

ID number	Date	Discharge Q (liters/s)	Amount of NaCl (g)
E01	6 December 2018	2.5	100
E02	11 December 2018	14.0	100
E03	8 January 2019	4.5	100
E04	11 January 2019	3.8	100
E05	23 January 2019	9.0	100
E06	24 January 2019	7.9	100
E07	28 January 2019	22.8	100
E08	4 February 2019	17.0	100
E09	5 February 2019	17.1	100
E10	8 February 2019	15.9	100
E11	25 February 2019a	5.3	100
E12	25 February 2019b	4.9	100
E13	25 February 2019c	4.7	100
E14	8 March 2019	28.6	100
E15	11 March 2019	25.2	100
E16	5 February 2020	75.0	100
E17	14 February 2020	36.7	250
E18	6 May 2020	1.3	150
E19	18 June 2020	0.9	100
E20	22 June 2020	0.4	100
E21	29 March 2021a	5.1	100
E22	29 March 2021b	5.0	100
E23	30 March 2021a	4.8	100
E24	30 March 2021b	4.7	100
E25	31 March 2021a	4.7	50
E26	31 March 2021b	4.6	100
E27	31 March 2021c	4.6	150
E28	31 March 2021d	4.1	200
E29	11 June 2021a	2.8	100
E30	11 June 2021b	2.5	100

and a slow-response to precipitation events. The fast-response is controlled by the surface runoff of event-water over the saturated organic soil in the riparian zone toward the stream channel (Antonelli et al., 2020a, 2020b; Bonanno et al., 2021; Wrede et al., 2015). The slow-response occurs when the amount of water from precipitation events exceeds the storage capacity at the hillslope. When this happens, the groundwater is laterally redistributed over the fractured bedrock from the hillslopes toward the stream channel causing an increase in discharge and a double-peak behavior in the hydrograph (Bonanno et al., 2021; Martínez-Carreras et al., 2016).

2.2. Tracer Experiments

We performed a total of 30 in-stream instantaneous tracer experiments between December 2018 and June 2021. For each experiment, we prepared a NaCl solution using 2 L of stream water and an amount of NaCl between 50 and 250 g (Table 2). We selected slug injections over constant-rate injections to minimize the influence of varying discharge on the BTC measurements (Ward, Gooseff, et al., 2013) and because they contain the same information as a constant-rate injection for conservative tracers (Payn et al., 2008). We injected the tracer solution in a turbulent pool at the beginning of the study reach, assuring complete mixing. We measured electric conductivity (EC) at the end of the investigated reach (55 m from the injection point) via a portable conductivity meter (WTW TetraCon 3310) providing a resolution of 0.1 µS/cm from 0 to 199 µS/cm and 1 µS/cm from 200 to 1,999 µS/cm (accuracy $\pm 0.5\%$ of the value and temperature automatically compensated). Calibration of EC to chloride concentration was conducted in the laboratory using a known volume of water sampled at the measurement location before the tracer injection. The conversion into chloride concentration was obtained via an EC-Cl⁻ regression line ($R^2 > 0.99$) and the background concentration was removed using water samples taken before and after the end of the experiment at the injection point. Discharge (Q [L³/T]) during each tracer injection was evaluated using the recorded BTC via the dilution gauging method (Beven et al., 1979; Butterworth et al., 2000). The experiments were carried out across different discharge stages of the Weierbach over three hydrologic years (October 2018-October 2021, Table 2; Figures 1c and 1d. BTC data available in Bonanno, Barnich, et al., 2022).

2.3. Water Table Measurements and Groundwater Flow Direction

The study site is equipped with a network of 36 wells and seven piezometers (Figure 1a). The wells are in the near-stream domain, while the piezometers are located in the streambed (Bonanno et al., 2021). We manually measured the water level in the monitoring network using a water level meter

(Eijkelkamp 11.03) immediately before the injection of the tracers into the stream channel. The piezometers in the stream channel allowed us to measure both the groundwater hydraulic head and water depth in the stream channel.

We obtained the groundwater table at the study site by interpolating the measured groundwater level in the network via spline interpolation (Matlab R2020a, The Mathworks, Natick, MA). The interpolated groundwater table was used to evaluate the groundwater flow direction in the area between the wells in the monitoring network during each instantaneous tracer injection. If the extracted flow direction was pointing away or toward the stream channel, then that stream section was considered in gaining and losing condition, respectively (Bonanno et al., 2021). The area with the groundwater flow direction pointing away from the stream channel and then returning to the stream channel was considered as the size of the hyporheic zone on the *xy* plane at the investigated hydrologic conditions. The corresponding volume of the hyporheic zone was approximated as equal to the hyporheic zone area on the *xy* plane multiplied by the average water depth in the stream channel. The volume of

the hyporheic zone so obtained was then averaged over the reach length (55 m), to estimate the average area of the hyporheic zone on the plane perpendicular to the stream reach. It is important to note that, compared to the hyporheic zone extension on the *xy* plane, the area of the hyporheic zone averaged on the reach length was not directly measured. However, this quantity is particularly valuable because it allows for a direct comparison with the area of the transient storage as measured by the TSM, since both of them are a measure of the transient storage zone area perpendicular to the flow direction (Runkel et al., 2003).

2.4. Evaluation of Streambed Micro-Topography

We conducted four Laser Imaging Detection and Ranging (LIDAR) scans at the study site via terrestrial Faro Photon 120/20 (accuracy ± 2 mm at 25 m) prior to the tracer experiments (20 November 2018 to evaluate the size and the spatial distribution of the fragmented slates over the fractured bedrock in the streambed. Each scan was analyzed via Cloud Compare (version 2.11.3, 2020). The stream reach has been subdivided into transects with a 1 cm resolution, perpendicular to the stream talweg and distanced 15 cm from each other (black lines, Figure 1b). We evaluated the size and distribution of the slate fragments above the talweg for each LIDAR stream channel section. Knowing the water level recorded in the stream channel, each transect has also been used to evaluate the length of the wetted perimeter for each water level (blue, green and red lines, Figure 1b). We used the calculated wetted perimeter and the successive evaluation of the wetted stream area via TSM to evaluate the average hydraulic radius (H_R) during each tracer injection.

2.5. Formulation of the Transient Storage Model

The formulation of the TSM reads (Bencala & Walters, 1983):

$$\begin{cases} \frac{\partial C}{\partial t} = -v \frac{\partial C}{\partial x} + \frac{1}{A} \frac{\partial}{\partial x} \left(AD \frac{\partial C}{\partial x} \right) + \alpha (C_{\rm TS} - C) \\ \frac{\partial C_{\rm TS}}{\partial t} = -\alpha \frac{A}{A_{\rm TS}} (C_{\rm TS} - C) \end{cases}$$
(1)

where t is time [T], x is the distance from the injection point along the stream reach [L], A [L²] is the stream discharge cross-sectional area, v [L/T] is the flow velocity, D [L²/T] is the longitudinal dispersion coefficient, C and C_{TS} are the concentration of the observed tracer in the stream channel and in the storage zone, respectively [M/L³], α [1/T] is the exchange coefficient between the stream channel and the storage zone and A_{TS} [L²] is the area of the transient storage zone. Three primary assumptions are associated with the formulation of the TSM (Harvey et al., 1996): (a) negligible transport in the hyporheic zone parallel to the stream, (b) exponential residence time distribution (RTD) in the transient storage zone, and (c) in-stream water storage and hyporheic exchange are jointly described by the transient storage parameters α and A_{TS} . This last limitation reflects the lumped nature of the single storage zone described in the TSM which is thus not able to distinguish between the cross-sectional area of in-stream storage zones and the hyporheic zone (Runkel et al., 2003).

2.6. Calibration and Identifiability of Transient Storage Model Parameters

2.6.1. Identifiable and Non-Identifiable TSM Parameters

TSM parameters are usually obtained via visual fitting or inverse modeling approaches, such as OTIS-P (Table 1). Despite the good model performances that can be obtained from inverse modeling approaches, the parameters might be non-identifiable (Kelleher et al., 2019) and may not capture the underlying processes well. Thus, several authors advocated the identification of a "behavioral" parameter population in TSMs via identifiability analysis (i.e., parameter sets satisfying certain performance thresholds) since this is a preferable and more informative outcome than a singular best set of parameter values (Beven, 2001; Kelleher et al., 2019; Wagener, Lees, & Wheater, 2002; Wagener et al., 2002, 2002; Wlostowski et al., 2013).

The "identifiability" term is used to indicate how certain a parameter is in a model application. If a good model performance occurs in only a relatively narrow parameter interval compared to the distribution of its possible values, then the parameter can be considered identifiable. On the contrary, if a good model performance is distributed across a relatively large parameter interval compared to the distribution of its possible values, then

Table 3

List of Calibration Parameters and Used Initial Ranges for the First Iteration of the GlaDy Identifiability Analysis

Abbrev.	Parameter	Lower bound	Upper bound
v [m/s]	Flow velocity	$0.5 \cdot v_{ADE}$	$2 \cdot v_{ADE}$
<i>A</i> [m ²]	Advective channel cross-sectional area	$0.5 \cdot A_{ADE}$	$2 \cdot A_{ADE}$
<i>D</i> [m ² /s]	Longitudinal dispersion coefficient	10-4	$2 \cdot D_{ADE}$
α [s ⁻¹]	Transient storage exchange rate	10 ⁻⁵	0.1
$A_{\rm TS}$ [m ²]	Transient storage cross-sectional area	10-5	1

Note. The subscript ADE indicates the best-fitting model parameter obtained after the BTC fitting using the advection-dispersion equation (cfr Section 2.6.2).

the parameter can be considered non-identifiable (Ward et al., 2017). In studies where the identifiability of TSM parameters has been investigated via random sampling approaches, α and $A_{\rm TS}$ have proved to be rarely identifiable (Kelleher et al., 2013; Ward, Gooseff, et al., 2013, 2017).

2.6.2. The GlaDy Identifiability Analysis for TSM

We obtained TSM parameters and their identifiability via a novel iterative modeling approach that combines random sampling with Global identifiability analysis and Dynamic identifiability analysis (GlaDy, Bonanno, 2022). The GlaDy identifiability analysis requires a certain number of parameter sets and the corresponding model performance compared to the observed BTC. By "parameter set" we refer to a specific combination of v, A, D, α , and $A_{\rm TS}$ parameter values, while by "parameter space" we refer to the range of a parameter between the selected lower and the upper bounds in the random parameter sampling. Model performance was evaluated with the

Nash-Sutcliffe objective function (NSE). We selected NSE because it allows consistency between the dynamic identifiability analysis and global identifiability analysis used in the iterative modeling approach (Bonanno, Blöschl, & Klaus, 2022; Wagener, Lees, & Wheater, 2002). Additionally, NSE is a normalized objective function that is beneficial for comparing performances across runs and among tracer experiments. Global identifiability analysis addresses the identifiability of the TSM parameters using as model performances the NSE evaluated on the entire BTC, while the dynamic identifiability analysis addresses the identifiability of the TSM parameters over time, meaning that identifiability is studied along a moving window over the BTC (Wagener, Lees, & Wheater, 2002).

Compared to other work addressing the identifiability of TSM we here considered as a calibration parameter also the flow velocity, v. This is because it was proven that keeping v as a constant model parameter might cause misestimation of the other TSM parameters during identifiability analysis (Bonanno, Blöschl, & Klaus, 2022). By considering v as a calibration parameter we also avoid the uncertainty inevitably bonded to experimental velocity measurements or the discharge evaluation via dilution gauging method, since $Q = v \cdot A$ (Schmadel et al., 2010).

Before the application of the TSM, we simulated every tracer experiment via the classic advection-dispersion equation (ADE) to avoid any initial assumptions of the advection-dispersion parameters (v, A, D) that might affect the results of the iterative modeling approach. 35,000 ADE parameter sets have been uniformly sampled from the feasible parameter space (see Bonanno, Blöschl, & Klaus, 2022) and the parameter set with the highest NSE defined the optimal ADE fitting with parameters v_{ADE} , A_{ADE} , D_{ADE} , and performance NSE_{ADE}.

Starting from ADE results (v_{ADE} , A_{ADE} , D_{ADE}) the GlaDy identifiability analysis conduces a first iteration by randomly sampling via Latin Hypercube sampling 35,000 TSM parameters set (v, A, D, α , and A_{TS}) over a relatively large parameter space (Table 3) and evaluating the corresponding model performance using OTIS as a solver. This number of parameter sets was chosen for every TSM iteration because 35,000 parameter sets were proven to have always less than 2% error in the mean and standard deviation of the top 10% results compared to 115,000 parameter sets, when the identifiability conditions are met (Bonanno, Blöschl, & Klaus, 2022). The upstream boundary condition was set as instantaneous (IBOUND = 1, in OTIS) over three time steps: at zero and 2 s, the concentration was set to zero, while at one second, the concentration was calculated as the mass of the solute divided by the product of the randomly sampled velocity and area, multiplied by the time interval used for the injection (1 s). As a result, each randomly sampled parameter set also considered its specific concentration as the upstream boundary condition.

The GlaDy identifiability analysis is iterative, meaning that a successive TSM iteration depends on the results of the identifiability analysis of the previous iteration. This is similar to previous studies that used a random sampling approach combined with behavioral thresholds (Kelleher et al., 2013; Ward, Kelleher, et al., 2017). The following TSM iterations rely on a constrained parameter space depending on the best-performing upper and lower bound obtained by the results of the global and dynamic identifiability analysis.

2.6.3. Conditions for Global and Dynamic Identifiability

Globally identifiable parameters have: (a) univocal peak of performance in parameter versus objective function plots (Ward et al., 2017); (b) cumulative distribution function (CDF) corresponding to the best 0.1% of the model results deviating from the 1:1 line and from parameter CDF corresponding to the best 10% of the model results (Kelleher et al., 2019); (c) the two-sample Kolmogorov-Smirnov (K-S) test indicating a statistically relevant difference in the CDF corresponding to the best 0.1% and 10% results ($p \le 0.05$, Bonanno, Blöschl, & Klaus, 2022):

$$[K, p] = \max|F(P_{0.1}) - F(P_{10})|$$
⁽²⁾

where $F(P_{0,1})$ and $F(P_{10})$ are the cumulative distribution function of a parameter *P* respectively for the best 0.1% and the best 10% of the model results. Dynamic identifiability analysis indicates the 90% confidence interval of a parameter compared to the considered parameter space over different sections of the BTC. The evaluation of one minus the width of the 90% confidence interval over the entire parameter range indicates the "information content" of a certain parameter over the BTC. Information content values close to one indicate stronger parameter identifiability in that investigated section of the BTC compared to lower information content values (Bonanno, Blöschl, & Klaus, 2022; Wagener, Lees, & Wheater, 2002).

2.6.4. Dependency of Identifiability of TSM Parameters With the Number of Iterations

The GlaDy identifiability analysis was finalized once all model parameters (v, A, D, α , and A_{TS}) satisfied the global identifiability conditions defined above in their selected parameter space. We repeated the sampling of 35,000 parameter sets over the parameter space that indicated identifiability until we obtained at least 1,000 parameter sets that perform better than the ADE (NSE > NSE_{ADE}). This was done to obtain a statistically relevant number of parameter sets with satisfactory model performances because only TSM parameter sets with better performances than the ADE should be used for the interpretation of model results (Bonanno, Blöschl, & Klaus, 2022). The successive analysis of the transient storage process and transport metrics were conducted only on the behavioral parameter sets (NSE > NSE_{ADE}).

After every model iteration, we evaluated the mean and standard deviation of the top 10% NSE for all the modeling results and behavioral parameter sets only. This is because the top 10% of the results are often used as a behavioral threshold in several studies addressing the identifiability of TSM parameters (Kelleher et al., 2019; Wagener, Lees, & Wheater, 2002; Ward, Kelleher, et al., 2017). Following Pianosi et al. (2015) and Bonanno, Blöschl, and Klaus (2022), we interpreted a decrease in the mean and standard deviation of NSE with an increasing number of TSM iterations as an increase in model identifiability. On the contrary, the constant mean and standard deviation of NSE with an increasing number of iterations have been interpreted as the model is unable to increase TSM performances with increasing iterations.

2.6.5. Comparison With OTIS-P Modeling Approach

We here compared our results to the more traditional inverse modeling approach, OTIS-P. This is because when OTIS-P is used, no identifiability study is generally conducted to address the reliability of the TSM parameters (Fabian et al., 2011; Jin & Ward, 2005; McKnight et al., 2004; Schmid et al., 2010) while only a few studies compared random sampling approach with OTIS-P (Ward et al., 2017, 2018). By comparing the GlaDy identifiability analysis introduced in Bonanno, Blöschl, and Klaus (2022) with OTIS-P we aim for a more robust assessment of parameter estimation and their identifiability not previously reported in the literature.

OTIS-P uses a non-linear regression scheme to minimize the residual sum of squares between the modeled BTC and the observed BTC and returns the 95% confidence interval for the estimated TSM parameters. Following Runkel (1998), we carried out multiple OTIS-P iterations starting from different initial parameter values $(A = A_{ADE}, D = D_{ADE} \text{ and } \alpha = 0.1 \text{ s}^{-1}, A_{TS} = 0.1 \text{ m}^2; \alpha = 0.01 \text{ s}^{-1}, A_{TS} = 0.001 \text{ s}^{-1}, A_{TS} = 0.001 \text{ m}^2)$. This was done to avoid false model convergence to a local minimum. We applied OTIS-P in consecutive steps, setting the results obtained from the previous modeling output as starting parameter values of the successive software application. We finalized the use of OTIS-P when parameter values changed less than 0.1% between subsequent runs (Ward et al., 2017). When OTIS-P was not able to converge to a unique set of parameter values or indicated convergence errors, we discarded its results.

2.7. Metrics Characterizing Solute Transport in Stream

We computed several metrics from the best-performing parameter sets (NSE > NSE_{ADE}) related to solute transport and storage in the study reach.

We evaluated the average residence time of a tracer molecule in the transient storage zone $(RT_s [T])$ and the average time a tracer molecule remains in the stream channel before passing into the storage zone $(RT_Q [T])$ (Runkel, 2002; Thackston & Schnelle, 1970):

R

$$T_Q = \frac{1}{\alpha} \tag{3}$$

$$\mathrm{RT}_{S} = \frac{A_{\mathrm{TS}}}{\alpha \cdot A} \tag{4}$$

We obtained the total water flux exchanged between the stream channel and the storage zone. This was done by multiplying the average water flux through the storage zone per unit length of the stream channel by the reach length $L(q_s [L^3/T], \text{modified from Harvey et al.}, 1996)$:

$$q_s = \alpha \cdot A \cdot L \tag{5}$$

The hydrological retention factor ($R_{\rm H}$ [T/L]) is a useful metric to compare transient storage among reaches and under different discharges. $R_{\rm H}$ quantifies the storage zone residence time of water per unit of stream reach traveled and it can be evaluated as (Morrice et al., 1997):

$$R_H = \frac{A_{TS}}{A \cdot v} \tag{6}$$

We evaluated F_{MED} [-] which incorporates the role of advective transport and transient storage processes (Runkel, 2002):

$$F_{\text{MED}} \cong \left(1 - e^{\left(-L\frac{\alpha}{v}\right)}\right) \frac{A_{\text{TS}}}{A_{\text{TS}} + A}$$
(7)

 F_{MED} is the fraction of median travel time due to transient storage (or percent if multiplied by 100). Increasing values of F_{MED} have to be interpreted as relatively larger importance of the transient storage processes on the solute transport in the stream corridor.

From the measured water depth in the stream channel and other streambed characteristics we evaluated the Darcy-Weisbach friction factor (f[-]). This quantity has been related to streambed complexity and in-stream storage zones and its formulation reads (Bencala & Walters, 1983; Hart et al., 1999; Thackston & Schnelle, 1970):

$$f = \frac{8g \cdot d \cdot S}{v^2} \tag{8}$$

where g [L/T²] is the gravitational constant, S [L/L] is the slope of the energy grade line estimated from the stream channel slope (Zarnetske et al., 2007), d [L] is the average water depth measured in the stream channel.

We evaluated Manning's roughness coefficient n [-] to assess whether an increase in transient storage area is linked to an increase of friction with in-stream sediments due to a larger contact area with the streambed fractured slate:

$$n = \frac{H_R^{\frac{2}{3}} \cdot S^{\frac{1}{2}}}{v} \tag{9}$$

3. Results

3.1. Transient Storage Model Parameters and Their Identifiability

The GlaDy identifiability analysis approach was effective in identifying the model parameters for the 30 tracer experiments regardless of the hydrologic conditions (Figure 2). The best-fitting parameter sets obtained at the end





Figure 2. Results of Global identifiability analysis and Dynamic identifiability analysis (GlaDy) identifiability analysis reported as parameter values plotted against the corresponding NSE values. Black and red dots indicate parameter sets and corresponding model performances with worse (black) and better (red) performance than the ADE. Green dots indicate calibration results via OTIS-P and the gray areas indicate the corresponding 95% parameter confidence limits. The blue triangles indicate the best-performing parameter set obtained via the GlaDy identifiability analysis. The horizontal red line indicates the adopted behavioral threshold (NSE_{ADE}). Results are reported for five different experiments with higher values of discharge: first row: E290 = 2.8 liters/s; Second row: E04 = 3.8 liters/s; Third row: E06 = 7.9 liters/s; fourth row: E10 = 15.9 liters/s; Fifth row: E14 = 28.6 liters/s.

of the TSM iterations outperformed in terms of objective function NSE the OTIS-P results for 20 of the 30 experiments (Table S1 in Supporting Information S1). OTIS-P also proved ineffective in calibrating the TSM parameters for three tracer experiments, due to convergence errors in the inverse modeling scheme. The stream discharge assessed through the GlaDy identifiability analysis exhibited values that closely aligned with those derived from the dilution gauging method. This was evident from the plot of the product of ν -A from the best-fitting parameter sets obtained via the GlaDy identifiability analysis against the discharge values obtained through the dilution gauging method, which displayed a 1:1 relationship (Figure S1 in Supporting Information S1).

The distribution of the model errors for the top 10% of model results with NSE > NSE_{ADE} indicates that higher discharge during an experiment is linked to a better performance of the TSM compared to experiments with lower discharge (boxplots, Figure 3). This is observable in the results from the iterative modeling approach and OTIS-P (blue triangles and green dots in Figure 3) and by the fact that the Spearman correlation coefficient (Rs) of NSE_{ADE} for increasing discharge was positive and significant (Rs = 3.996, with *p*-value < 0.05). The difference in performance between the TSM results and the NSE_{ADE} is smaller for experiments with higher discharge compared to experiments with low discharge (red squares, Figure 3), however this difference was not significantly correlated with discharge (p-value > 0.05). Our results also show that after four or five TSM iterations the mean and standard deviation of the model errors for the top 10% of model results and for the top 10% of model results with NSE > NSE_{ADF} are constant with the increasing number of iterations (Figure 4). This outcome shows that the high number of iterations only matters for obtaining the best-fitting parameter sets (number of red dots in Figure 2 and blue triangles in Figures 2 and 3) but does not control the distribution of the behavioral parameter sets. This is because model performances for the top 10% of model outcomes converge toward unique values after a few TSM iterations and do not decrease considerably with the number of iterations (Figure 4). The GlaDy identifiability analysis also shows that under low discharge there is a sharp decrease in model errors with the number of TSM iterations, and that the behavioral threshold of obtaining at least 1,000 parameter sets with NSE > NSE_{ADE} is satisfied after a few iterations (<10 iterations, blue lines Figure 4). On the contrary, under higher discharge, the





Figure 3. Left *y*-axis: Boxplot of the distributions of model error for the 10% best-performing parameter sets (NSE > NSE_{ADE}). Blue triangles and green dots indicate model performances obtained via the Global identifiability analysis and Dynamic identifiability analysis (GlaDy) identifiability analysis and OTIS-P respectively. The bottom *x*-axis reports the ID code of the tracer experiments, while the upper *x*-axis indicates the corresponding discharge conditions [liters/s]. Right *y*-axis: the red squares indicate the difference between NSE_{ADE} and the NSE of the best-performing parameter sets via the iterative modeling approach. Red crosses on the bottom *x*-axis indicated ineffective application of OTIS-P (false convergence).

GlaDy identifiability analysis shows a smoother decrease of model performances with the increasing number of model iterations, and the behavioral threshold of obtaining at least 1,000 parameter sets with NSE > NSE_{ADE} is satisfied only after a large number of iterations (>10 iterations, red lines Figure 4).

3.2. Extension and Contraction of the Hyporheic Zone and Development of In-Stream Storage Zones

To aid the interpretation of the TSM results we evaluated the size of the hyporheic zone during each tracer experiment by calculating the area of the near-stream groundwater pointing away and then returning toward the stream channel using the near-stream groundwater monitoring well network (cfr. Section 2.3). Groundwater table



Figure 4. Dependency of (a) mean and (b) standard deviation of model error for the top 10% of the modeling results on the number of TSM iterations. Results have been shown for three experiments (E29 = 2.78 liters/s; E12 = 4.88 liters/s; E08 = 16.97 liters/s). Each TSM iteration includes model performances for 35,000 parameter sets. An increase in TSM iterations has to be interpreted as an increase in the total number of parameter sets considered for the evaluation of mean and standard deviation of NSE values (e.g., 10 TSM iterations include model results for 350,000 parameter sets).





Figure 5. Contraction and extension of the size of the hyporheic zone exemplified for two experiments with (a) 75.0 liters/s (E16) and (b) 0.4 liters/s (E20). The blue dots in (a) and (b) indicate the location of the groundwater monitoring well network used to measure the groundwater table during each tracer injection. The colored areas in (a) and (b) indicate the groundwater flow direction normalized with the respect to the direction of the stream channel on the *xy* plane (flow direction of 0° equal to -72° on the *xy* plane). Green areas indicate groundwater flow direction flowing parallel to the stream channel; blue areas indicate groundwater pointing toward the stream channel. The dashed yellow areas indicate the area of the hyporheic zone receiving stream water, while the dashed blue areas indicate the area of the hyporheic zone returning water to the stream channel. Together, the yellow and the blue dashed area indicate the area of the hyporheic zone on the *xy* plane. The black line indicates the perimeter of the maximum size of the hyporheic zone conditional to the well network. (c) Orange dots, dependency of the evaluated size of the hyporheic zone on the *xy* plane with discharge. The subplot (green dots) reports the same dependency, but indicated via the percentage area of the hyporheic zone on the plane perpendicular to the stream reach with discharge. The red and blue curves indicate the best-fitting relationships for values below and after 5 liters/s.

observations and the groundwater flow direction indicate that the hyporheic zone is smaller when discharge is higher (Figure 5). The size of the hyporheic zone on the *xy* plane varies between 29.79 m² at 75 liters/s (E16, Figures 5a) and 226.34 m² at 0.4 liters/s (E20, Figure 5b) which are 6.4% and 48.8% of the maximum size of the hyporheic zone (black perimeter, Figures 5a and 5b). The size of the hyporheic zone significantly decreases under higher discharge (Figure 5c, Rs = -0.90, *p-value* < 0.01). Despite the size of the hyporheic zone on the *xy* plane decreases with discharge, the water depth in the stream channel has a logarithmic trend with the discharge ($R^2 = 0.81$, *p-value* < 0.01, and Rs = 0.85, *p-value* < 0.01, plot not shown). As a result, the volume of the hyporheic zone displays an increase with discharge for values below ~5 liters/s and a decrease with discharge for values above ~5 liters/s (Figure 5d). The normalization of the volume of the hyporheic zone over the reach length allowed us to evaluate the change in the average size of the area of the hyporheic zone perpendicular to the stream reach with varying discharge conditions (Figure 5e). This relationship was well approximated by a logarithmic increase ($R^2 = 0.919$, *p-value* < 0.01, red line in Figure 5e) for discharge values lower than 5 liters/s, and by a power law ($R^2 = 8.24$, *p-value* < 0.01, blue line in Figure 5e) for discharge values above than 5 liters/s.





Figure 6. Evaluation of (a) hydraulic radius H_{R^*} (b) Darcy-Weisbach friction factor *f*, (c) Manning's roughness coefficient *n*, and percentage of the stream reach where the (d) hydraulic radius or the (e) stream water level was higher than the 5-percentile, average, 95-percentile and maximum height of the slate fragment that makes up the streambed against discharge for the 30 tracer experiments. Horizontal black lines indicate the 5- and 95-percentile limits of discharge for the top 10% TSM results with NSE > NSE_{ADE}. Vertical black lines indicate the 5 and 95 percentile limits of $H_{R^*}f$, and *n* evaluated for the total LIDAR transects.

The hydraulic radius H_R decreases for discharge between 0 and 5 liters/s and increases with discharge for values above 5 liters/s (Figure 6a). The Darcy-Weisbach friction factor f significantly decreases with higher discharge (Figure 6b, Rs = -0.885, *p*-value < 0.01) and it can be approximated by a power law fitting function ($R^2 = 0.706$, *p*-value < 0.01). Similarly, Manning's roughness coefficient n significantly decreases with higher discharge (Figure 6c, Rs = -0.872, *p-value* < 0.01) and this trend is well approximated by an exponential function ($R^2 = 0.977$, *p*-value < 0.01). The size of the slate fragments above the talweg evaluated for each LIDAR transect allows us to extrapolate how many stream transects (evaluated as a % over the total) have a water table and hydraulic radius above a certain height of the slate fragments (reported as 5-percentile, average, 95-percentile, and a maximum of the slate height, Figures 6d and 6e). Our results show that the hydraulic radius H_R is above the 5-percentile, the average, and the 95-percentile of the fragment height for the majority of the stream reach (95%) when discharge is above 7.9, 15.9, and 75 liters/s respectively (Figure 6d). Similarly, the measured stream water level is above the 5-percentile, the average, the 95-percentile, and the maximum size of the height of the fragmented slates in most of the stream reach (95%) when discharge is above 1.3, 2.5, 7.9, and 14 liters/s, respectively.

3.3. How Does Transient Storage Change Between Experiments?

Our results from the GlaDy identifiability analysis show that advection-dispersion parameters increase with discharge (Figures 7a, 7c and 7e). The increase of v with discharge is significant (Rs = 0.945, p-value < 0.01) and follows a linear and power law function ($R^2 = 0.971$, $R^2 = 0.975$, respectively; p-value < 0.01). The increase of A and D with discharge is significant (Rs = 0.932 and Rs = 0.833, respectively, with p-value < 0.01) following a power law increase ($R^2 = 0.937$ and $R^2 = 0.862$, for A and D respectively; p-value < 0.01). The TSM parameters also increase with discharge. α increases significantly with discharge (Rs = 0.784, p-value < 0.01, Figure 7b) following a quadratic fitting function ($R^2 = 0.936$, p-value < 0.01). A_{TS} shows high variability for discharge stages lower than 5 liters/s having a non-significant correlation with discharge (Rs = -0.437, p-value > 0.1). However, for discharge values larger than 5 liters/s the A_{TS}

parameter significantly increases with discharge (Rs = 0.815, p-value < 0.01) following a power law fitting function ($R^2 = 0.895$, p-value < 0.01). The correlation of A_{TS} with discharge results in a similar behavior of A_{TS}/A (Figure 7f). The ratio A_{TS}/A shows a sharp decrease with discharge for values lower than ~5 liters/s (power law $R^2 = 0.915$, p-value < 0.01, with Rs = -0.657, p-value = 0.012), and increases with discharge for values above 5 liters/s ($R^2 = 0.566$, p-value < 0.01 using a power law function with Rs = 0.588, p-value = 0.018).

The transport metrics show different patterns against discharge (Figure 8). The total water flux exchanged between the stream channel and the storage zone (q_s) significantly increases with discharge (Rs = 0.866, p-value < 0.01) and this relationship was well approximated by both a power law and a cubic function ($R^2 > 0.87$, p-value < 0.01, Figure 8a). The hydrologic retention factor ($R_{\rm H}$) is significant and negatively correlated with increasing discharge (Rs = -0.872, p-value < 0.01, Figure 8b) following an exponential or a power law decrease function ($R^2 > 0.96$, p-value < 0.001). The average residence time in the stream channel ($RT_{\rm Q}$) and the transient storage zone ($RT_{\rm s}$) are negatively and significantly correlated with discharge (Rs = -0.784 and Rs = -0.857, p-value < 0.01, Figures 8c and 8d respectively). The decrease of $RT_{\rm Q}$ and $RT_{\rm s}$ with Q can be well approximated by a power law ($R^2 = 0.627$ and $R^2 = 0.932$ respectively, p-value < 0.01). The fraction of median travel time due to transient storage shows a decreasing and an increasing trend with increasing discharge, for values respectively below and above 5 liters/s (Figure 8 e, f) When the discharge is comprised between 0 and 5 liters/s, $F_{\rm MED}$ decreases significantly with discharge (Rs = -0.684, p-value < 0.01) following a linear ($R^2 = 0.776$, p-value < 0.01) or a power law (0.851, p-value < 0.01) decrease. However, when the discharge is above 5 liters/s, $F_{\rm MED}$ increases significantly with



Figure 7. TSM parameters in relation to discharge during the experiments. Plots are reported for (a) flow velocity, (c) advective channel cross-sectional area, (e) longitudinal dispersion coefficient, (b) transient storage exchange rate, (d) transient storage cross-sectional area, and (f) ratio between transient storage cross-sectional area and advective channel cross-sectional area. Dots show the best parameter set obtained from the iterative modeling approach. Vertical and horizontal black lines indicate 5 and 95 percentile limits of the top 10% results with NSE > NSE_{ADE}. Gradient colors indicate the percentage of the near-stream groundwater receiving and returning stream water to the stream channel (cfr. Figure 5).

discharge (Rs = 0.691, *p*-value < 0.01) following a linear ($R^2 = 0.558$, *p*-value < 0.01) or a power law ($R^2 = 0.567$, *p*-value < 0.01) increase.

4. Discussion

4.1. Parameter Identifiability in the Transient Storage Model Depends on Discharge During the Tracer Experiments

When TSM parameters are non-identifiable they are interdependent and a change of a certain parameter would be balanced by a proportional change of other parameters, eventually leading to the same model performances (Camacho & González, 2008; Kelleher et al., 2019; Wagener, Lees, & Wheater, 2002; Wlostowski et al., 2013). Compared to the available literature, no previous study directly addressed the role of increasing discharge on the identifiability of TSM parameters. Our results indicate that the interdependency of TSM parameters increases with higher discharge during the experiment. This is visible from the less pronounced peak of performances of v, A, and D (red dots, Figure 2) and the fact that the space of the advection-dispersion parameters showing satisfactory model performances (NSE > NSE_{ADE}) increases with the discharge (red dots, Figure 2). This results in a





Figure 8. Same as Figure 7, but reporting (a) the total water flux exchanged between the stream channel and the storage zone, (b) the hydrological retention factor, (c) the average residence time of a tracer molecule in the stream channel, and (d) in the transient storage zone transport metrics as a function of discharge. The last two plots report the fraction of median travel time due to transient storage respectively for discharge conditions (e) before and (f) after 5 liters/s.

larger 5- and 95-percentile limits of the top 10% TSM results compared to experiments with low discharge, where advection-dispersion parameters show a rather narrow peak of performances (Figure 2, Table S1 in Supporting Information S1; black lines, Figure 7).

Our outcomes indicate that the advection-dispersion parameters explain largely the shape of BTC during high discharge and that the transient storage process added to the advection-dispersion equation contributes to a modest improvement of model performances compared to the ADE. Since larger portions of the BTC can be explained by the advection-dispersion process it may be argued that the transient storage parameters are less identifiable because the transient storage process is less important with increasing discharge. This interpretation is corroborated by the obtained significant correlation between NSE_{ADE} with increasing discharge and the limited increase of the TSM performances compared to the ADE performances with higher discharge (red squares, Figure 3).

The observed parameter interaction together with the larger number of TSM iterations and sampled parameter sets needed to obtain identifiable TSM parameters at higher discharge conditions (Figure 4) support the interpretation that greater parameter interactions cause poorer parameter identifiability and model performances (Kelleher et al., 2013; Ward, Payn, et al., 2013). This study positions itself alongside other studies that investigate the identifiability of TSM parameters and show that the advection-dispersion process becomes predominant over

the transient storage processes during higher discharge. For example, Wagner and Harvey (1997) were the first to point toward the role of the advection-dispersion process for the identifiability of the transient storage parameters while Kelleher et al. (2013) and Bonanno, Blöschl, and Klaus (2022) demonstrated advection-dispersion parameters control progressively larger portions of the BTC under higher discharge, while transient-storage parameters control progressively larger portions of the BTC under lower discharge.

The results reported in this study present a novel viewpoint and a potential elucidation for the non-identifiability reported in prior investigations, where the efficacy of employing random sampling to estimate TSM parameters has been inconsistent, with no clear understanding of the underlying reasons (see Table 1). We believe that the advection-dispersion process was not predominant at study sites with relatively low discharge meaning that α and $A_{\rm TS}$ were explanatory of large portions of the BTC. As a result, the TSM improved substantially performances compared to ADE due to the pronounced tail of the BTC, and the identifiability of TSM parameters was achieved via two iterative random sampling for a total of 100,000 parameter sets (Ward et al., 2018; first BTC in Ward, Kelleher, et al., 2017). On the other hand, headwater reaches characterized by steep channel gradients and relatively short investigated reaches indicated non-identifiability of TSM parameters (Kelleher et al., 2013; Ward, Payn, et al., 2013; second and third BTCs in Ward, Kelleher, et al., 2017). This is probably because the advection-dispersion process at these study sites dominated the tracer transport and the investigated parameter space and/or the used number of parameter sets (often $\leq 100,000$) did not allow to target NSE > NSE_{ADE} for a sufficient number of parameter sets to show identifiability.

The GlaDy identifiability analysis introduced in Bonanno, Blöschl, and Klaus (2022) and used in this work can also demonstrate that selecting a narrow (< two orders of magnitude) parameter interval in a random-sampling approach can cause an "apparent" non-identifiability in TSM. If we had sampled a parameter from a narrow space around the peak of performance (e.g., α between 0.001 and 0.003 s⁻¹ for experiment E06, results not shown), the identifiability analysis results would lead us to the conclusion that the parameter was non-identifiable. However, this same interval shows optimal performances when a wider parameter space is sampled (Figure 2). This may account for the lack of parameter identifiability observed in previous studies that have focused on a narrow range of TSM parameters (Camacho & González, 2008; Wagener, Lees, & Wheater, 2002; Wlostowski et al., 2013), as a limited parameter space may not permit a clear improvement in model performance. Consequently, the identifiability of TSM parameters could remain hidden, especially if an insufficient number of parameter sets are sampled (<100,000, Bonanno, Blöschl, & Klaus, 2022; Ward et al., 2017). Our findings demonstrate that the GlaDy identifiability analysis can address parameter identifiability across a parameter space spanning several orders of magnitude, which is consistent with recent recommendations for identifiability analysis (Pianosi et al., 2016) and prior research (Kelleher et al., 2013; Kelleher et al., 2019; Ward, Kelleher, et al., 2017; Ward, Payn, et al., 2013).

Our OTIS-P simulations demonstrated satisfactory model performance, as the calibrated TSM parameters were within the same parameter space as those obtained from the GlaDy identifiability analysis (Figure 2; SI). While these outcomes suggest that OTIS-P can yield robust results, past work highlighted that OTIS-P does not provide information about the identifiability and performances of TSM parameters across their feasible range, making it challenging to determine if OTIS-P results shall be used for process interpretation or not (Kelleher et al., 2013; Knapp & Kelleher, 2020; Ward, Kelleher, et al., 2017). Moreover, the best-fitting parameter sets obtained from the GlaDy identifiability analysis outperformed in terms of objective function (NSE) the OTIS-P results for most tracer experiments and allowed us to obtain parameter identifiability even for BTCs where OTIS-P failed to converge (Table S1 in Supporting Information S1). Furthermore, it is important to note that while GlaDy identifiability analysis demonstrated superior performance compared to OTIS-P for 20 of the 30 experiments, the difference in terms of NSE objective function was negligible for many of them (Table S1 in Supporting Information S1). Therefore, for future studies investigating TSM it may be advisable to use random sampling over a parameter space defined as a neighborhood range of TSM parameters obtained via OTIS-P. However, in cases where OTIS-P fails to converge or when TSM parameters cannot be identified through classic random-sampling approaches, the GlaDy identifiability analysis has been demonstrated to be a powerful and flexible tool for achieving systematic parameter identifiability across a wide range of hydrologic conditions. This represents a significant and previously unreported achievement in TSMs.

The performances of OTIS-P and the GlaDy identifiability analysis reported in this study are certainly typical of the study site. Future research is necessary to evaluate the applicability of the GlaDy identifiability analysis to

other stream reaches, as tracer experiments conducted in different geomorphological settings have been shown to strongly influence the shape of the BTC and TSM performances (D'Angelo et al., 1993; Edwardson et al., 2003; Hall et al., 2002; Zarnetske et al., 2007). Additionally, the ability to obtain parameter identifiability may have been facilitated by the simplistic formulation of the TSM, which treats transient storage as a single storage area with a solute residence time that follows an exponential decay. Future work should test the GlaDy identifiability analysis with modifications to TSM that increase the number of parameters, thus their interaction and the potential for non-identifiability (Knapp & Kelleher, 2020). Among the diverse TSM formulations aimed at providing a more realistic representation of solute transport in streams, the GlaDy identifiability analysis should be applied to TSMs with multiple transient storage areas (Choi et al., 2000; Fabian et al., 2011), TSMs with different residence time distribution laws (Bottacin-Busolin & Marion, 2010; Marion et al., 2008; Gooseff et al., 2005; Haggerty et al., 2002), and BTCs of non-conservative solutes (Kelleher et al., 2019).

4.2. Dynamics of Transient Storage Processes Under Different Hydrologic Conditions

The higher longitudinal dispersion coefficients *D* in experiments with higher discharge are also in line with the TSM formulation and with the observed increase of the Reynolds number with discharge (between 2.99·10³ and 2.22·10⁵, linear and quadratic function fit with $R^2 > 0.99$, *p-value* < 0.01, plots not shown). *D* is responsible for the longitudinal spreading of the tracer above and behind the center of the solute pulse, thus it is expected to increase with increasing streambed roughness and channel complexity (Gooseff, Bencala, et al., 2008). Furthermore, *D* is proportional to the spatial variability of the flow velocity across the velocity field, thus it increases with discharge as a result of increases in wetted stream area and the consequent spatial heterogeneity in the velocity profile. This is despite increasing turbulence with discharge would theoretically slightly decrease *D* (Fischer et al., 1979). Similarly, many experimental studies report that *D* is proportional to discharge (Kashefipour & Falconer, 2002).

The trend of F_{MED} for discharge below 5 liters/s (Figure 8e) together with the large size of the hyporheic area and volume assessed through groundwater measurements (Figures 5c–5e) suggest a non-negligible contribution of hyporheic exchange to transient storage for experiments with low discharge (<5 liters/s) compared to experiments with higher discharge. This interpretation is also supported by the A_{TS}/A ratio for discharge below 5 liters/s (Figure 7f), the high values of the hydrologic retention factor R_{H} (>1 s/m Figure 8b), and the negative correlation of RT_{s} and RT_{o} with discharge (Figures 8c and 8d).

This result is consistent with previous investigations at the study sites reporting low near-stream groundwater table as a result of a discontinued hillslope-riparian-stream connectivity during dry hydrologic conditions (Bonanno et al., 2021; Figure 9a). Our TSM results are consistent with previous studies where gradients from the stream channel toward the adjacent groundwater have been linked to hyporheic transport (González-Pinzón et al., 2015; Harvey & Bencala, 1993; Kasahara & Wondzell, 2003). However, our findings also indicate a significant role of in-stream transient storage during low discharge (<5 liters/s). This can be deduced from the high values of the friction factor *f* and the roughness coefficient *n* obtained for experiments with low discharge and the fact that the hydraulic radius H_R was at its minimum at $Q \sim 5$ liters/s. These results indicate that the wetted perimeter increases more than the wet stream area for higher discharge between 0 and ~5 liters/s. This in turn causes a greater proportion of the streambed material to be submerged, but not completely (Figures 6d and 6e). This partial submergence of larger areas of the streambed material causes the development of secondary flowpaths among the slate fragments and turbulences in the shaded area immediately downstream causing an increase in the in-stream transient storage. Our results show that transient storage during experiments under low discharge at the study site (Q < 5 liters/s) cannot be explained by hyporheic exchange or in-stream transient storage alone, but as a combination of both (Figure 9c).

Higher discharge at the study site is characterized by an increase in gradients from the adjacent groundwater toward the stream channel, indicating a persistent hillslope-stream connectivity on the west and the east hillslopes (Bonanno et al., 2021). This is consistent with our results indicating a decrease in the size of the groundwater area and volume receiving water from the stream channel (Figures 4c and 4d) for experiments with discharge higher than 5 liters/s. These results show that the hyporheic exchange decreases with higher discharge suggesting that transient storage is mainly controlled by in-stream transient storage. However, the observed high percentage of slate fragments in the stream channel that are entirely submerged below the water table with higher discharge (Figures 6d and 6e) indicates that the secondary flowpaths and recirculation zones controlling in-stream transient





Figure 9. Hydrograph of the Weierbach catchment for four hydrologic years considering (a) the increase of the streambed sediments completely submerged in the advective stream channel and (b) the increase and decrease of the hyporheic zone area perpendicular to the stream channel. (c-e) Figure reports a perceptual model of the mechanisms controlling the transient storage in the investigated stream reach with increasing discharge conditions.

storage at lower discharge are now part of the advective stream channel (Figure 9e). Also, the observed trend of the hydraulic radius with discharge shows that the wet area increases more than the wet perimeter for discharge higher than 5 liters/s (Figures 6a-6c). These results provide additional evidence that also in-stream transient storage becomes less important for solute transport with higher discharge at the study site.

As stream discharge increases further, the streambed sediments become completely submerged below the in-stream water level (Q > 17 liters/s, Figure 6e) and the hyporheic zone area and volume progressively decrease (Figures 5c-5e). Also, the decrease of the roughness n and friction factor f (Figures 6b and 6c) indicates a decreasing shear velocity u^* with increasing discharge. This result suggests a lower contribution of shear stress to in-stream transient storage controlled by turbulences ($u^* \propto gdS$ as in Equation 8, see Fischer et al., 1979). Instead, transient storage at the study site might be ruled by increasing spatial heterogeneities of the velocity gradients due to the increasing wetted area with higher discharge, by the occurrence of recirculating zones above submerged cavities and slates in the streambed (Jackson et al., 2013), or by short-lasting hyporheic exchange controlled by local pressure gradients of the stream water on the streambed (Cardenas & Wilson, 2007). This interpretation aligns with the trend observed in the evaluated F_{MED} and RT_{s} metrics, which indicate that transient storage becomes increasingly relevant with higher discharge, but lasting only a few seconds within the stream corridor (Figures 8d and 8f). This outcome is consistent with research at other sites characterized by low hyporheic exchange, where higher discharge during tracer experiments resulted in relatively lower in-stream transient storage compared to lower discharge (Martí et al., 1997; Zarnetske et al., 2011). However, our results on $F_{\rm MED}$ demonstrate that an increase in discharge does not necessarily indicate an absence of in-stream storage (Figure 8f). Instead, our outcomes suggest that the underlying mechanisms controlling in-stream storage change with increasing discharge and shift from secondary flowpaths and recirculating zones due to partially submerged streambed materials (Figures 5e and 9d) to increasing velocity gradients in larger wetted areas and recirculating eddies in the water column probably due to submerged streambed cavities (Gooseff, Payn, et al., 2008; Jackson et al., 2013).

Compared to the extension of the hyporheic zone area and volume (Figures 4c and 4d), the deduced order of magnitude of A_{TS} and q_S (Figures 7d and 8a) indicates that we were likely unable to capture longer flowpaths and residence time of the stream water into large areas of the hyporheic zone as evaluated via the groundwater measurements. This is particularly evident when the area of the hyporheic zone perpendicular to the stream channel (Figure 5e) is compared to the TSM parameter A_{TS} (Figure 7d). Our results show that A_{TS} underestimates the area of the hyporheic zone at low discharge conditions and especially around 5 liters/s. However, with increasing discharge the A_{TS} parameter increases and approaches the value of the hyporheic zone area experimentally measured and averaged on the reach length (Figure 5e). This result highlights the limitations of using tracer experiments alone to capture the magnitude of the hyporheic zone and exchange, particularly at lower discharge conditions. This is because while previous studies reported instantaneous injections to be capable of returning model information comparable to that of continuous injections for conservative tracers (Gooseff, Payn, et al., 2008; Payn et al., 2008), other studies also highlight instantaneous injections are limited by the available "window of detection", which is biased toward faster transient storage processes and shallow hyporheic exchange (Harvey & Wagner, 2000; Jin & Ward, 2005; Wondzell, 2006).

Our study is the first study to our knowledge that is addressing the concurrence of different processes controlling the transient storage via the use of TSM parameters, the near-stream groundwater levels, and the streambed micro-topography under several hydrologic conditions. We recognize that the adopted strategy is not without criticism. As an example, the groundwater monitoring well network is not designed to capture pressure gradients at the surface water–streambed interface that is recognized to be a non-negligible source of hyporheic flowpaths with increasing turbulence and discharge (Cardenas & Wilson, 2007; Packman & Bencala, 2000). Also, the evaluation of the hyporheic zone on the *xy* plane is limited by the available number of wells at the study site, meaning that it could potentially be larger than the one we estimated, especially at low discharge conditions (Figures 5a and 5c). In addition, the LIDAR scans might not be representative of the streambed micro-topography and slate distribution above the talweg across the relatively long investigated period (from December 2018 to June 2021) and more scans could have provided more robust results. Despite some limitations, our approach bypassed the "window of detection" problem typical of tracer injections and assessed the dynamic role of hyporheic exchange and in-stream transient storage on water transport across the hydrologic year.

4.3. A Dynamic Perceptual Model of Transient Storage Processes

We acknowledge that our TSM results and interpretation are likely representatives of the studied reach only. However, our findings underscore the importance of investigating a wide range of hydrologic conditions to establish a robust understanding of the processes controlling transient storage in stream corridors. We used the obtained relationships between increasing discharge with the transient storage area perpendicular to the stream channel (Figure 5e) and the amount of streambed sediments completely submerged in the advective stream channel (Figure 6e) to derive the potential role of hyporheic extension and in-stream transient storage across 4 years in the Weierbach catchment (Figures 9a and 9b). Due to the obtained relationships between TSM parameters and discharge, these plots are indicative of the expected role of in-stream and hyporheic storage processes occurring at the study sites for a large spectrum of hydrologic conditions and in absence of tracer experiments. The results collected in this work also allowed us to build a perceptual model for a qualitative representation of the mechanisms controlling the transient storage at the study site under varying discharge (Figures 9c-9e). This is a step forward in one of the current research frontiers in hydrology, namely the spatial and temporal development of the interfaces between the stream water and the adjacent groundwater (Blöschl et al., 2019), which was argued to be caused by limited experimental measurements in the stream corridor (Ward & Packman, 2019), parameters non-identifiability (Knapp & Kelleher, 2020), and the low amount of investigated hydrologic conditions (Ward, 2016).

Under low discharge conditions (Q < 5 liters/s) the extension of the hyporheic zone on the xy plane is large (Figures 5b and 5c). However, the low water level in the stream channel does not allow the development of a relevant gradient between the in-stream water level and the adjacent groundwater, resulting in a small hyporheic volume and area of hyporheic zone perpendicular to the stream reach (Figures 5d and 5e, 9b). Also, the majority of the streambed sediments are only partially submerged (Figures 6d and 6e an 9a), which allows the development of multiple flowpaths in the stream channel in an almost-dry streambed (Figure 9a). These mechanisms have a relatively high impact on transient storage, as also indicated by the increasing F_{MED} and the trend of the other transport metrics (Figure 8). With the progressive increase of discharge, the wet perimeter increases more than the wetted area until ~ 5 liters/s (Figure 6a) and many in-stream sediments get submerged, but not completely (Figures 6d and 6e and 9a). At this stage, the groundwater level adjacent to the stream channel is low enough to allow a large portion of the stream reach to be in losing condition toward the groundwater (Figures 5b and 5c), while the increasing water level in the stream channel control the development of increasing volumes of hyporheic zones (Figures 5d and 5e). These hydrologic conditions create a stage where there is a large potential for in-stream recirculating zones and hyporheic exchange, but where the overall transient storage effect obtained by TSM is lower compared to lower hydrologic conditions (Figures 7f, 8b, 8c, and 9d), probably as a result of the "window of detection" problem typical of instantaneous tracer injections. With increasing wetness conditions the wet area of the stream channel increases more than the wet perimeter (Figure 6a) and in-stream sediments get completely submerged in the stream reach at \sim 17 liters/s (Figures 6e and 9a). At this stage, the inflow from the adjacent hillslopes causes the predominance of near-stream groundwater gradients pointing toward the stream allowing only an occasional development of hyporheic exchange (Figure 5a) and a decreasing volume and area of hyporheic zone perpendicular to the stream channel (Figures 5c-5e). As a result, transient storage might be controlled by increasing velocity gradients in larger wetted areas, recirculating zones due to submerged cavities (Jackson et al., 2013) and by short-lasting hyporheic exchange controlled by pressure gradients that the monitoring well network at the study site is unable to capture (Cardenas & Wilson, 2007; Packman & Bencala, 2000).

This perceptual model obtained and the results reported in this work can help to understand why increasing discharge in past studies was related both to an increase and a decrease of α and $A_{\rm TS}$ parameters and apparent contradictory model interpretation (Table 1). Our study highlights the coexistence of many simultaneous mechanisms responsible for transient storage processes and that they can both increase and decrease with increasing discharge as a result of the complex interaction between in-stream water level, groundwater level, in-stream sediments, and recirculating zones. As a result, depending on a site-specific threshold for stream water level above and below streambed material, the transient storage parameters might have shown both an increase or a decrease with increasing discharge (D'Angelo et al., 1993; Edwardson et al., 2003; Fabian et al., 2011; Martí et al., 1997; Morrice et al., 1997), similar to our outcomes for discharge values below or above 5 liters/s. TSM results in past literature have also been linked to the occurrence of hyporheic exchange, and the observed trend of transient storage parameters with the streamflow discharge has been linked to a larger or lower importance of the hyporheic exchange in the stream corridor (Gooseff et al., 2003; Schmid et al., 2010; Valett et al., 1996;

Zarnetske et al., 2007). In our study, we proved that increasing transient storage parameters with discharge were both linked to a decrease and a decrease in the hyporheic exchange, highlighting the pivotal role of having near-stream groundwater measurement for a correct interpretation of TSM parameters. Eventually, we recognize that the spatial and temporal scale of the processes investigated at our study site is not representative of transient storage processes occurring in a larger-order stream and alluvial sites where streambed sediments are probably always completely submerged and other mechanisms could be predominant, such as meandering exchange (Boano et al., 2006), the occurrence of bars and dunes (Stonedahl et al., 2013), fine-sediment accumulation (Ward et al., 2018), and transient storage in large ponds in the stream channel (Hall et al., 2002; Magliozzi et al., 2018; Morrice et al., 1997).

We acknowledge that obtaining high-resolution data such as spatially dense groundwater monitoring and LIDAR scans may not be feasible for many research sites. However, our findings highlight the potential pitfalls of relying solely on phenomenological models, such as TSM, for interpreting transient storage processes. To ensure accurate characterization of hyporheic exchange, future studies should include measurements of water table gradients between the stream water and adjacent groundwater, which can be obtained through the use of relatively inexpensive shallow piezometers (Fabian et al., 2011; Nowinski et al., 2012). This would provide critical insight into whether a hyporheic exchange is occurring or if it is merely speculative. Additionally, although dry streambed microtopography may not always be available, manual sampling of in-stream streambed sediments and comparison of their size with in-stream water levels during tracer injections can still be valuable information for determining whether all streambeds have the potential to develop secondary flow paths and in-stream recirculating zones, or whether they are completely submerged within the advective stream channel.

5. Conclusion

Answering how and why transient storage processes change with different hydrologic conditions can bring a comprehensive assessment of their spatial and temporal role in regulating water quality in stream networks. In this study, we used an iterative modeling approach (the GlaDy identifiability analysis) to obtain identifiable TSM parameters for 30 tracer breakthrough experiments at different discharges in a headwater stream reach. We combined the model results, groundwater table observations, and measurements of streambed micro-topography to support the interpretation of different processes concurring to the transient storage of stream water through a wide range of hydrological conditions. Our work showed that the parameter space where advection-dispersion parameters were identifiable was wider under higher discharge, thus increasing the parameter interaction in the TSM. Our outcomes can thus explain the lack of parameter identifiability in several previous TSM studies and open up new challenges to address parameter identifiability in other model formulations implemented with many transient storage areas or with different residence time distributions. The introduction of the streambed micro-topography and groundwater table measurement provided valuable data for interpreting the TSM parameters and the transport metrics describing solute transport at the study site. Our model results showed that hyporheic exchange and in-stream transient storage control the transient storage of stream water during low discharge. Under higher discharge, the hyporheic zone becomes progressively less important in controlling transient storage, which is rather driven by in-stream storage probably controlled by increasing velocity gradients, pressure gradients on the streambed interface, and recirculating zones by submerged cavities. Our findings inspired the development of a perceptual model for a qualitative understanding of the processes governing transient storage at the study site. The relationships between submerged sediments and the hyporheic zone development with increasing discharge were also utilized to generate a hydrograph for the study site indicating the dynamic development of in-stream and hyporheic storage over a 4-year period. This outcome provides insights into the expected influence of transient storage processes prior to tracer experiments. Future work should combine tracer injections with information on streambed micro-topography and groundwater measurements in stream reaches characterized by different morphologies and hydrologic regimes. The combination of derived patterns between discharge, hyporheic area, stream water elevation, and streambed microtopography in several study sites would be key for understanding the spatial and temporal variation of transient storage processes in stream networks and across scales.

Data Availability Statement

The GlaDy identifiability analysis has been introduced by Bonanno, Blöschl, and Klaus (2022) and it is available in Bonanno (2022). Data associated with this manuscript are available in Bonanno, Barnich, et al. (2022).

Acknowledgments

This work was supported by funding from the Luxembourg National Research Fund (FNR) for doctoral training (PRIDE15/10623093/HYDRO-CSI). We would like to acknowledge the financial support of the Austrian Science Fund (FWF) as part of the Vienna Doctoral Programme on Water Resource Systems (DK W1219-N28). We thank the University of Trier (Department IV -Erdbeobachtung und Klimaprozesse) for borrowing the laser scanner Faro Photon 120 and Christian Bossung for helping in conducting the scanners and for providing the. BIN files of the scans. The authors would like to thank one anonymous reviewer, Dr. Robert Runkel, Dr. Fulvio Boano and the WRR's associate editor for their very constructive comments that substantially improved the manuscript quality. We thank Jean-François Iffly, Jérôme Juilleret, and Laurent Gourdol for their help in the installation of the monitoring well network. We thank Ginevra Fabiani, and Laurent Pfister for their fruitful input and discussions.

References

- Antonelli, M., Glaser, B., Teuling, A. J., Klaus, J., & Pfister, L. (2020a). Saturated areas through the lens: 1. Spatio-Temporal variability of surface saturation documented through thermal infrared imagery. *Hydrological Processes*, 34(6), 1310–1332. https://doi.org/10.1002/hyp.13698
- Antonelli, M., Glaser, B., Teuling, A. J., Klaus, J., & Pfister, L. (2020b). Saturated areas through the lens: 2. Spatio-Temporal variability of streamflow generation and its relationship with surface saturation. *Hydrological Processes*, 34(6), 1333–1349. https://doi.org/10.1002/hyp.13607
- Argerich, A., Martí, E., Sabater, F., Haggerty, R., & Ribot, M. (2011). Influence of transient storage on stream nutrient uptake based on substrata manipulation. Aquatic Sciences, 73(3), 365–376. https://doi.org/10.1007/s00027-011-0184-9
- Bencala, K. E., Gooseff, M. N., & Kimball, B. A. (2011). Rethinking hyporheic flow and transient storage to advance understanding of stream-catchment connections. *Water Resources Research*, 47(3), 1–9. https://doi.org/10.1029/2010WR010066
- Bencala, K. E., McKnight, D. M., & Zellweger, G. W. (1990). Characterization of transport in an acidic and metal-rich mountain stream based on a lithium tracer injection and simulations of transient storage. *Water Resources Research*, 26(5), 989–1000. https://doi.org/10.1029/ WR026i005p00989
- Bencala, K. E., & Walters, R. A. (1983). Simulation of solute transport in a mountain pool-and-riffle stream: A transient storage model. Water Resources Research, 19(3), 718–724. https://doi.org/10.1029/WR019i003p00718
- Beven*, K. (2001). How far can we go in distributed hydrological modelling? Hydrology and Earth System Sciences, 5(1), 1–12. https://doi.org/10.5194/hess-5-1-2001
- Beven, K., Gilman, K., & Newson, M. (1979). Flow and flow routing in upland channel networks. *Hydrological Sciences Journal*, 24(3), 303–325. https://doi.org/10.1080/02626667909491869
- Blöschl, G., Bierkens, M. F. P., Chambel, A., Cudennec, C., Destouni, G., Fiori, A., et al. (2019). Twenty-three unsolved problems in hydrology (UPH)–a community perspective. *Hydrological Sciences Journal*, 64(10), 1141–1158. https://doi.org/10.1080/02626667.2019.1620507
- Boano, F., Camporeale, C., Revelli, R., & Ridolfi, L. (2006). Sinuosity-driven hyporheic exchange in meandering rivers. *Geophysical Research Letters*, 33(18), 1–4. https://doi.org/10.1029/2006GL027630
- Boano, F., Harvey, J. W., Marion, A., Packman, A. I., Revelli, R., Ridolfi, R., & Wörman, A. (2014). Hyporheic flow and transport processes: Mechanisms, models, and bioghemical implications. *Reviews of Geophysics*, 52(4), 603–679. https://doi.org/10.1002/2012RG000417. Received
- Bonanno, E. (2022). BTC_analysis: GLaDY GLobal and DYynamic identifiability analysis BTC application (SoluteTransport). Zenodo. https://doi.org/10.5281/zenodo.7381262
- Bonanno, E., Barnich, F., Gourdol, L., Iffly, J. F., Juilleret, J., Pfister, L., & Klaus, J. (2022). HYDRO-CSI, Project 1.2: In-stream hydrology. Part 2: Instantaneous injections (version 1.0.0). [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.6457709
- Bonanno, E., Blöschl, G., & Klaus, J. (2021). Flow directions of stream-groundwater exchange in a headwater catchment during the hydrologic year. *Hydrological Processes*, 35(8), 1–18. https://doi.org/10.1002/hyp.14310
- Bonanno, E., Blöschl, G., & Klaus, J. (2022). Exploring tracer information in a small stream to improve parameter identifiability and enhance the process interpretation in transient storage models. *Hydrology and Earth System Sciences*, 26(23), 6003–6028. https://doi.org/10.5194/ hess-26-6003-2022
- Bottacin-Busolin, A., & Marion, A. (2010). Combined role of advective pumping and mechanical dispersion on time scales of bed form-induced hyporheic exchange. Water Resources Research, *46*, 1–12. https://doi.org/10.1029/2009WR008892
- Boulton, A. J., Datry, T., Kasahara, T., Mutz, M., & Stanford, J. A. (2010). Ecology and management of the hyporheic zone: Stream–groundwater interactions of running waters and their floodplains. *Journal of the North American Benthological Society*, 29(1), 26–40. https://doi. org/10.1899/08-017.1
- Butterworth, J. A., Hewitt, E. J., & McCartney, M. P. (2000). Discharge measurement using portable dilution gauging Flowmeters. *Water and Environment Journal*, *14*(6), 436–441. https://doi.org/10.1111/j.1747-6593.2000.tb00291.x
- Butturini, A., & Sabater, F. (1999). Importance of transient storage zones for ammonium and phosphate retention in a sandy-bottom Mediterranean stream. *Freshwater Biology*, *41*(3), 593–603. https://doi.org/10.1046/j.1365-2427.1999.00406.x
- Camacho, L. A., & González, R. A. (2008). Calibration and predictive ability analysis of longitudinal solute transport models in mountain streams. Environmental Fluid Mechanics, 8(5–6), 597–604. https://doi.org/10.1007/s10652-008-9109-0
- Cardenas, M. B., & Wilson, J. L. (2007). Exchange across a sediment-water interface with ambient groundwater discharge. Journal of Hydrology, 346(3–4), 69–80. https://doi.org/10.1016/j.jhydrol.2007.08.019
- Choi, J., Harvey, J. W., & Conklin, M. H. (2000). Characterizing multiple timescales and storage zone interaction that affect solute fate and transport in stream. Water Resources Research, 36(6), 1511–1518. https://doi.org/10.1029/2000wr900051
- D'Angelo, D. J., Webster, J. R., Gregory, S. V., & Meyer, J. L. (1993). Transient storage in Appalachian and cascade mountain streams as related to hydraulic characteristics. *Journal of the North American Benthological Society*, 12(3), 223–235. https://doi.org/10.2307/1467457
- Dudley-Southern, M., & Binley, A. (2015). Temporal responses of groundwater-surface water exchange to successive storm events. Water Resources Research, 51(2), 1112–1126. https://doi.org/10.1002/2014WR016623
- Edwardson, K. J., Bowden, W. B., Dahm, C., & Morrice, J. (2003). The hydraulic characteristics and geochemistry of hyporheic and parafluvial zones in Arctic tundra streams, North Slope, Alaska. *Advances in Water Resources*, 26(9), 907–923. https://doi.org/10.1016/S0309-1708(03)00078-2
- Fabian, M. W., Endreny, T. A., Bottacin-Busolin, A., & Lautz, L. K. (2011). Seasonal variation in cascade-driven hyporheic exchange, northern Honduras. *Hydrological Processes*, 25(10), 1630–1646. https://doi.org/10.1002/hyp.7924
- Fabiani, G., Schoppach, R., Penna, D., & Klaus, J. (2021). Transpiration patterns and water use strategies of beech and oak trees along a hillslope. *Ecohydrology*, 15(2), 1–18. https://doi.org/10.1002/eco.2382
- Fischer, H., Imberger, J., List, E., Koh, R., & Brooks, N. (1979). Mixing in inland and coastal waters. Academic Press.
- Glaser, B., Antonelli, M., Hopp, L., & Klaus, J. (2020). Intra-catchment variability of surface saturation insights from physically based simulations in comparison with biweekly thermal infrared image observations. *Hydrology and Earth System Sciences*, 24(3), 1393–1413. https://doi.org/10.5194/hess-24-1393-2020
- Glaser, B., Klaus, J., Frei, S., Frentress, J., Pfister, L., & Hopp, L. (2016). On the value of surface saturated area dynamics mapped with thermal infrared imagery for modeling the hillslope-riparian-stream continuum. *Water Resources Research*, 52(10), 8317–8342. https://doi. org/10.1002/2015WR018414
- González-pinzón, R., Ward, A. S., Hatch, C. E., Wlostowski, A. N., Singha, K., Gooseff, M. N., et al. (2015). A field comparison of multiple techniques to quantify groundwater Surface-water interactions. *Freshwater Science*, 34(August 2014), 139–160. https://doi.org/10.1086/679738
- González-Pinzón, R., Ward, A. S., Hatch, C. E., Wlostowski, A. N., Singha, K., Gooseff, M. N., et al. (2015). A field comparison of multiple techniques to quantify groundwater-surface-water interactions. *Freshwater Science*, 34(1), 139–160. https://doi.org/10.1086/679738

- Gooseff, M. N., Bencala, K. E., Wondzell, S. M., Service, U. F., Northwest, P., & Sciences, O. F. (2008). Solute transport along stream and river networks. In S. P. Rice, A. G. Roy, & B. L. Rhoads (Eds.), *River confluences, tributaries and the fluvial network*. John Wiley & Sons, Ltd. Retrieved from https://onlinelibrary.wiley.com/doi/book/10.1002/9780470760383
- Gooseff, M. N., Briggs, M. A., Bencala, K. E., McGlynn, B. L., & Scott, D. T. (2013). Do transient storage parameters directly scale in longer, combined stream reaches? Reach length dependence of transient storage interpretations. *Journal of Hydrology*, 483, 16–25. https://doi. org/10.1016/j.jhydrol.2012.12.046
- Gooseff, M. N., LaNier, J., Haggerty, R., & Kokkeler, K. (2005). Determining in-channel (dead zone) transient storage by comparing solute transport in a bedrock channel-alluvial channel sequence, Oregon. Water Resources Research, 41(6), 1–7. https://doi.org/10.1029/2004WR003513
- Gooseff, M. N., Payn, R. A., Zarnetske, J. P., Bowden, W. B., McNamara, J. P., & Bradford, J. H. (2008). Comparison of in-channel mobile-immobile zone exchange during instantaneous and constant rate stream tracer additions: Implications for design and interpretation of non-conservative tracer experiments. *Journal of Hydrology*, 357(1–2), 112–124. https://doi.org/10.1016/j.jhydrol.2008.05.006
- Gooseff, M. N., Wondzell, S. M., Haggerty, R., & Anderson, J. (2003). Comparing transient storage modeling and residence time distribution (RTD) analysis in geomorphically varied reaches in the Lookout Creek basin, Oregon, USA. Advances in Water Resources, 26(9), 925–937. https://doi.org/10.1016/S0309-1708(03)00105-2
- Haggerty, R., Wondzell, S. M., & Johnson, M. A. (2002). Power-law residence time distribution in the hyporheic zone of a 2nd-order mountain stream. *Geophysical Research Letters*, 29(13), 18-1–18–4. https://doi.org/10.1029/2002GL014743
- Hall, R. J. O., Bernhardt, E. S., & Likens, G. E. (2002). Relating nutrient uptake with transient storage in forested mountain streams. *Limnology* & Oceanography, 47(1), 255–265. https://doi.org/10.4319/lo.2002.47.1.0255
- Hart, D. D., Johnson, T. E., Bushaw-Newton, K. L., Horwitz, R. J., Bednarek, A. T., Charles, D. F., et al. (2002). Dam removal: Challenges and opportunities for ecological research and river restoration. *BioScience*, 52(8), 669–681. https://doi.org/10.1641/0006-3568(2002)052[06 69:DRCAOF]2.0.CO;2
- Hart, D. R. (1995). Parameter estimation and stochastic interpretation of the transient storage model for solute transport in streams. Water Resources Research, 31(2), 323–328. https://doi.org/10.1029/94WR02739
- Hart, D. R., Mulholland, P. J., Marzolf, E. R., DeAngelis, D. L., & Hendricks, S. P. (1999). Relationships between hydraulic parameters in a small stream under varying flow and seasonal conditions. *Hydrological Processes*, 13(10), 1497–1510. https://doi.org/10.1002/ (SICI)1099-1085(199907)13:10<1497::AID-HYP825>3.0.CO;2-1
- Harvey, J. W., & Bencala, K. E. (1993). The Effect of streambed topography on surface-subsurface water exchange in mountain catchments. Water Resources Research, 29(1), 89–98. https://doi.org/10.1029/92WR01960
- Harvey, J. W., & Wagner, B. J. (2000). Quantifying hydrologic interactions between streams and their subsurface hyporheic zones. In Streams and ground waters (pp. 3–44). Elsevier. https://doi.org/10.1016/B978-012389845-6/50002-8
- Harvey, J. W., Wagner, B. J., & Bencala, K. E. (1996). Evaluating the reliability of the stream tracer approach to characterize stream subsurface water exchange. *Water Resources Research*.
- Hissler, C., Martínez-Carreras, N., Barnich, F., Gourdol, L., Iffly, J. F., Juilleret, J., et al. (2021). The Weierbach experimental catchment in Luxembourg: A decade of critical zone monitoring in a temperate forest - from hydrological investigations to ecohydrological perspectives. *Hydrological Processes*, 35(5), 1–7. https://doi.org/10.1002/hyp.14140
- Jackson, T. R., Haggerty, R., & Apte, S. V. (2013). A fluid-mechanics based classification scheme for surface transient storage in riverine environments: Quantitatively separating surface from hyporheic transient storage. *Hydrology and Earth System Sciences*, 17(7), 2747–2779. https:// doi.org/10.5194/hess-17-2747-2013
- Jin, H. S., & Ward, G. M. (2005). Hydraulic characteristics of a small Coastal Plain stream of the southeastern United States: Effects of hydrology and season. *Hydrological Processes*, 19(20), 4147–4160. https://doi.org/10.1002/hyp.5878
- Karwan, D. L., & Saiers, J. E. (2009). Influences of seasonal flow regime on the fate and transport of fine particles and a dissolved solute in a New England stream. *Water Resources Research*, 45(11), 1–9. https://doi.org/10.1029/2009WR008077
- Kasahara, T., & Wondzell, S. M. (2003). Geomorphic controls on hyporheic exchange flow in mountain streams. Water Resources Research, 39(1), SBH3-1–SBH3-14. https://doi.org/10.1029/2002wr001386
- Kashefipour, S. M., & Falconer, R. A. (2002). Longitudinal dispersion coefficients in natural channels. Water Research, 36, 1596–1608. https:// doi.org/10.1016/s0043-1354(01)00351-7
- Kelleher, C., Wagener, T., McGlynn, B., Ward, A. S., Gooseff, M. N., & Payn, R. A. (2013). Identifiability of transient storage model parameters along a mountain stream. *Water Resources Research*, 49(9), 5290–5306. https://doi.org/10.1002/wrcr.20413
- Kelleher, C., Ward, A., Knapp, J. L. A., Blaen, P. J., Kurz, M. J., Drummond, J. D., et al. (2019). Exploring tracer information and model framework trade-offs to improve estimation of stream transient storage processes. *Water Resources Research*, 55(4), 3481–3501. https://doi. org/10.1029/2018WR023585
- Klaus, J., & Jackson, C. R. (2018). Interflow is not binary: A continuous shallow perched layer does not imply continuous connectivity. Water Resources Research, 54(9), 5921–5932. https://doi.org/10.1029/2018WR022920
- Knapp, J. L. A., & Kelleher, C. (2020). A perspective on the future of transient storage modeling: Let's stop chasing our tails. Water Resources Research, 56(3), 1–7. https://doi.org/10.1029/2019WR026257
- Krause, S., Lewandowski, J., Grimm, N. B., Hannah, D. M., Pinay, G., McDonald, K., et al. (2017). Ecohydrological interfaces as hot spots of ecosystem processes. *Water Resources Research*, 53(8), 6359–6376. https://doi.org/10.1002/2016WR019516
- Lees, M. J., Camacho, L. A., & Chapra, S. (2000). On the relationship of transient storage and aggregated dead zone models of longitudinal solute transport in streams. Water Resources Research, 36(1), 213–224. https://doi.org/10.1029/1999WR900265
- Legrand-Marcq, C., & Laudelout, H. (1985). Longitudinal dispersion in a forest stream. Journal of Hydrology, 78(3-4), 317-324. https://doi.org/10.1016/0022-1694(85)90109-X
- Magliozzi, C., Grabowski, R. C., Packman, A. I., & Krause, S. (2018). Toward a conceptual framework of hyporheic exchange across spatial scales. *Hydrology and Earth System Sciences*, 22(12), 6163–6185. https://doi.org/10.5194/hess-22-6163-2018
- Marion, A., Zaramella, M., & Bottacin-Busolin, A. (2008). Solute transport in rivers with multiple storage zones: The STIR model. Water Resources Research, 44(10), W10406. https://doi.org/10.1029/2008WR007037
- Martí, E., Grimm, N. B., Fisher, S. G., Marti, E., Grimm, N. B., & Fisher, S. G. (1997). Pre- and post-flood retention efficiency of nitrogen in a Sonoran Desert stream. Journal of the North American Benthological Society, 16(4), 805–819. https://doi.org/10.2307/1468173
- Martínez-Carreras, N., Hissler, C., Gourdol, L., Klaus, J., Juilleret, J., Iffly, J. F., & Pfister, L. (2016). Storage controls on the generation of double peak hydrographs in a forested headwater catchment. *Journal of Hydrology*, 543, 255–269. https://doi.org/10.1016/j.jhydrol.2016.10.004
- Mason, S. J. K., McGlynn, B. L., & Poole, G. C. (2012). Hydrologic response to channel reconfiguration on Silver Bow Creek, Montana. Journal of Hydrology, 438–439, 125–136. https://doi.org/10.1016/j.jhydrol.2012.03.010

- McKnight, D. M., Runkel, R. L., Tate, C. M., Duff, J. H., & Moorhead, D. L. (2004). Inorganic N and P dynamics of Antarctic glacial meltwater streams as controlled by hyporheic exchange and benthic autotrophic communities. *Journal of the North American Benthological Society*, 23(2), 171–188. https://doi.org/10.1899/0887-3593(2004)023<0171:INAPDO>2.0.CO;2
- Montgomery, D. R., & Buffington, J. M. (1997). Channel-reach morphology in mountain drainage basins. Geological Society of America Bulletin, 109(5), 596–611. https://doi.org/10.1130/0016-7606(1997)109<0596:CRMIMD>2.3.CO;2
- Morrice, J. A., Valett, H. M., Dahm, C. N., & Campana, M. E. (1997). Alluvial characteristics, groundwater–surface water exchange and hydrological retention in headwater streams. *Hydrological Processes*, 11(3), 253–267. https://doi.org/10.1002/ (SICI)1099-1085(19970315)11:3<253::AID-HYP439>3.0.CO;2-J
- Moser, D. P., Fredrickson, J. K., Geist, D. R., Arntzen, E. V., Peacock, A. D., Li, S. M. W., et al. (2003). Biogeochemical processes and microbial characteristics across groundwater-surface water boundaries of the Hanford Reach of the Columbia river. *Environmental Science and Technol*ogy, 37(22), 5127–5134. https://doi.org/10.1021/es034457v
- Nordin, C. F. J., & Troutman, B. M. (1980). Longitudinal dispersion in rivers: The persistence of skewness in observed data. Water Resources Research, 16(1), 123–128. https://doi.org/10.1029/WR016i001p00123
- Nowinski, J. D., Cardenas, M. B., Lightbody, A. F., Swanson, T. E., & Sawyer, A. H. (2012). Hydraulic and thermal response of groundwater-surface water exchange to flooding in an experimental aquifer. *Journal of Hydrology*, 472–473, 184–192. https://doi.org/10.1016/j.jhydrol.2012.09.018
- Packman, A. I., & Bencala, K. E. (2000). Modeling surface–subsurface hydrological interactions. In J. B. Jones & P. J. Mulholland (Eds.), Streams and ground waters (pp. 45–80). Elsevier. https://doi.org/10.1016/b978-012389845-6/50003-x
- Payn, R. A., Gooseff, M. N., Benson, D. A., Cirpka, O. A., Zarnetske, J. P., Bowden, W. B., et al. (2008). Comparison of instantaneous and constant-rate stream tracer experiments through non-parametric analysis of residence time distributions. *Water Resources Research*, 44(6), 1–10. https://doi.org/10.1029/2007WR006274
- Pianosi, F., Keith, B., Freer, J., Hall, J. W., Rougier, J., Stephenson, D. B., & Wagener, T. (2016). Sensitivity analysis of environmental models: A systematic review with practical workflow. *Environmental Modelling & Software*, 79, 214–232. https://doi.org/10.1016/j.envsoft.2016.02.008
- Pianosi, F., Sarrazin, F., & Wagener, T. (2015). A Matlab toolbox for global sensitivity analysis. *Environmental Modelling & Software*, 70, 80–85. https://doi.org/10.1016/j.envsoft.2015.04.009
- Rodriguez, N. B., & Klaus, J. (2019). Catchment travel times from composite StorAge Selection functions representing the superposition of streamflow generation processes. Water Resources Research, 55(11), 9292–9314. https://doi.org/10.1029/2019WR024973
- Runkel, R. L. (1998). One-dimensional transport with inflow and storage (OTIS): A solute transport model for streams and rivers (Vol. 98). U.S. Geol. Surv. Water Resour. Invest. Rep. Denver. University of Michigan Library. https://doi.org/10.1002/wrcr.20277-bib-0038
- Runkel, R. L. (2002). A new metric for determining the importance of transient storage. Journal of the North American Benthological Society, 21(4), 529–543. https://doi.org/10.2307/1468428
- Runkel, R. L., McKnight, D. M., & Rajaram, H. (2003). Modeling hyporheic zone processes. Advances in Water Resources, 26(9), 901–905. https://doi.org/10.1016/s0309-1708(03)00079-4
- Schmadel, N. M., Neilson, B. T., & Stevens, D. K. (2010). Approaches to estimate uncertainty in longitudinal channel water balances. *Journal of Hydrology*, 394(3–4), 357–369. https://doi.org/10.1016/j.jhydrol.2010.09.011
- Schmadel, N. M., Ward, A. S., Kurz, M. J., Fleckenstein, J. H., Zarnetske, J. P., Hannah, D. M., et al. (2016). Stream solute tracer timescales changing with discharge and reach length confound process interpretation. *Water Resources Research*, 52(4), 3227–3245. https://doi. org/10.1002/2015WR018062
- Schmid, B. H., Innocenti, I., & Sanfilippo, U. (2010). Characterizing solute transport with transient storage across a range of flow rates: The evidence of repeated tracer experiments in Austrian and Italian streams. *Advances in Water Resources*, 33(11), 1340–1346. https://doi.org/10.1016/j.advwatres.2010.06.001
- Scott, D. T., Gooseff, M. N., Bencala, K. E., & Runkel, R. L. (2003). Automated calibration of a stream solute transport model: Implications for interpretation of biogeochemical parameters. *Journal of the North American Benthological Society*, 22(4), 492–510. https://doi. org/10.2307/1468348
- Smith, J. W. N. (2005). Groundwater surface water interactions in the hyporheic zone (2005th ed.). Environment Agency. Retrieved from http:// www.environment-agency.gov.uk/
- Stanford, J. A., & Ward, J. (1988). The hyporheic habitat of river ecosystems. Nature, 335(6185), 64–66. https://doi.org/10.1038/332141a0
- Stonedahl, S. H., Harvey, J. W., & Packman, A. I. (2013). Interactions between hyporheic flow produced by stream meanders, bars, and dunes. Water Resources Research, 49(9), 5450–5461. https://doi.org/10.1002/wrcr.20400
- Thackston, E. L., & Schnelle, K. (1970). Predicting effects of dead zones on stream mixing. Journal of the Sanitary Engineering Division American Society of Civil Engineers, 93(2), 319–331. https://doi.org/10.1061/jsedai.0001078
- Valett, H. M., Morrice, J. A., Dahm, C. N., & Campana, M. E. (1996). Parent lithology, surface-groundwater exchange, and nitrate retention in headwater streams. *Limnology & Oceanography*, 41(2), 333–345. https://doi.org/10.4319/lo.1996.41.2.0333
- Voltz, T., Gooseff, M., Ward, A. S., Singha, K., Fitzgerald, M., & Wagener, T. (2013). Riparian hydraulic gradient and stream-groundwater exchange dynamics in steep headwater valleys. *Journal of Geophysical Research: Earth Surface*, 118(2), 953–969. https://doi.org/10.1002/ jgrf.20074
- Wagener, T., Camacho, L. A., & Wheater, H. S. (2002). Dynamic identifiability analysis of the transient storage model for solute transport in rivers (pp. 199–211).
- Wagener, T., Lees, M. J., & Wheater, H. S. (2002). A toolkit for the development and application of parsimonious hydrological models (Vol. 2, p. 34). Water Resources Publications.
- Wagner, B. J., & Harvey, J. W. (1997). Experimental design for estimating parameters of rate-limited mass transfer: Analysis of stream tracer studies. Water Resources Research, 33(7), 1731–1741. https://doi.org/10.1029/97WR01067
- Ward, A. S. (2016). The evolution and state of interdisciplinary hyporheic research. Wiley Interdisciplinary Reviews: Water, 3(1), 83–103. https:// doi.org/10.1002/wat2.1120
- Ward, A. S., Gooseff, M. N., Voltz, T. J., Fitzgerald, M., Singha, K., & Zarnetske, J. P. (2013). How does rapidly changing discharge during storm events affect transient storage and channel water balance in a headwater mountain stream? *Water Resources Research*, 49(9), 5473–5486. https://doi.org/10.1002/wrcr.20434
- Ward, A. S., Kelleher, C. A., Mason, S. J. K., Wagener, T., McIntyre, N., McGlynn, B., et al. (2017). A software tool to assess uncertainty in transient-storage model parameters using Monte Carlo simulations. *Freshwater Science*, 36(1), 195–217. https://doi.org/10.1086/690444
- Ward, A. S., Morgan, J. A., White, J. R., & Royer, T. V. (2018). Streambed restoration to remove fine sediment alters reach-scale transient storage in a low-gradient fifth-order river, Indiana, USA. *Hydrological Processes*, 32(12), 1786–1800. https://doi.org/10.1002/hyp.11518
- Ward, A. S., & Packman, A. I. (2019). Advancing our predictive understanding of river corridor exchange. WIREs Water, 6(1), e1327. https://doi.org/10.1002/wat2.1327

- Ward, A. S., Payn, R. A., Gooseff, M. N., McGlynn, B. L., Bencala, K. E., Kelleher, C. A., et al. (2013). Variations in surface water-ground water interactions along a headwater mountain stream: Comparisons between transient storage and water balance analyses. *Water Resources Research*, 49(6), 3359–3374. https://doi.org/10.1002/wrcr.20148
- Ward, A. S., Wondzell, S. M., Schmadel, N. M., Herzog, S., Zarnetske, J. P., Baranov, V., et al. (2019). Spatial and temporal variation in river corridor exchange across a 5th-order mountain stream network. *Hydrology and Earth System Sciences*, 23(12), 5199–5225. https://doi. org/10.5194/hess-23-5199-2019
- Wlostowski, A. N., Gooseff, M. N., Bowden, W. B., & Wollheim, W. M. (2017). Stream tracer breakthrough curve decomposition into mass fractions: A simple framework to analyze and compare conservative solute transport processes. *Limnology and Oceanography: Methods*, 15(2), 140–153. https://doi.org/10.1002/lom3.10148
- Wlostowski, A. N., Gooseff, M. N., & Wagener, T. (2013). Influence of constant rate versus slug injection experiment type on parameter identifiability in a 1-D transient storage model for stream solute transport. *Water Resources Research*, 49(2), 1184–1188. https://doi.org/10.1002/ wrcr.20103
- Wondzell, S. M. (2006). Effect of morphology and discharge on hyporheic exchange flows in two small streams in the Cascade Mountains of Oregon, USA. *Hydrological Processes*, 20(2), 267–287. https://doi.org/10.1002/hyp.5902
- Wondzell, S. M. (2011). The role of the hyporheic zone across stream networks. *Hydrological Processes*, 25(22), 3525–3532. https://doi. org/10.1002/hyp.8119
- Wörman, A. (1998). Analytical solution and timescale for transport of reacting solutes in rivers and streams. Water Resources Research, 34(10), 2703–2716. https://doi.org/10.1029/98wr01338
- Wrede, S., Fenicia, F., Martínez-Carreras, N., Juilleret, J., Hissler, C., Krein, A., et al. (2015). Towards more systematic perceptual model development: A case study using 3 Luxembourgish catchments. *Hydrological Processes*, 29(12), 2731–2750. https://doi.org/10.1002/hyp.10393
- Zarnetske, J. P., Gooseff, M. N., Brosten, T. R., Bradford, J. H., McNamara, J. P., & Bowden, W. B. (2007). Transient storage as a function of geomorphology, discharge, and permafrost active layer conditions in Arctic tundra streams. *Water Resources Research*, 43(7), 1–13. https:// doi.org/10.1029/2005WR004816
- Zarnetske, J. P., Haggerty, R., Wondzell, S. M., & Baker, M. A. (2011). Dynamics of nitrate production and removal as a function of residence time in the hyporheic zone. *Journal of Geophysical Research*, *116*(1), 1–12. https://doi.org/10.1029/2010JG001356